UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K/A

Amendment No. 1

Transition Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 Commission file number 001-31539 SMENERGY COMPANY (Exact name of registrant as specified in its charter) Delaware (Sittle or other jurisdiction (I.R.S. Employer Identification No.) of incorporation or organization) 1775 Sherman Street, Stint 1200, Devere, Colorado (Address of principal executive offices) (Registrant's telephone number, including area code) Securities registered pursuant to Section 12(b) of the Act: Title of each tass Common stock, S.D par value Securities registered pursuant to Section 12(b) of the Act: Name of end exchange on which registered Securities registered pursuant to Section 12(b) of the Act: Name of end exchange on which registered Indicate by check mark if the registrant is a well-known seasoned laster, as defined in Rule 405 of the Securities Act. Yes No. Indicate by check mark whether the registrant (1) has filed all reports pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to the short period to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to this such reports, and (2) has been subject to such filing requirements for the past 90 days. Yes 320 No. Indicate by check mark whether the registrant was required to the shorter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post ad number to Rule 465 of Regulation S-1 (§ 232.495 of this chapter) is not contained been and post ad number to Rule 465 of Regulation S-1 (§ 232.495 of this chapter) is not contained herein, and will not be contained to the best of registrant's knowledge, in definitive prexy or information statements incorporated by reference in Part III of this Form 10-K or any unendment to this Form 10-K. S.D. Indicate by check mark whether	☒ Annual Report Pursuant to Section 13 or 15(d) of the	Securities Exchange Act of 1934
Transition Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 Commission file number 001-31539 SM ENERGY COMPANY (Exact name of registrant as specified in its charrer) Delaware (State or other jurisdiction of incorporation or organization) 1775 Sherman Street, Suite 1200, Denver, Colorado (Address of principal executive offices) Cuil 38 51-8140 (Registrant's telephone number, including area code) Securities registered pursuant to Section 12(b) of the Act. Title of each class Common stock, \$01 par value Securities registered pursuant to Section 12(g) of the Act. None Indicate by check mark if the registrant is a well-known seusoned issuer, as defined in Rule 405 of the Securities Act Yes Do D Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 15(d) of the Act. Yes Do D Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 15(d) of the Act. Yes Do D Indicate by check mark whether the registrant (1) has submitted electronically and posted pursuant to Rule 405 of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes IS No D Indicate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes IS No D Indicate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes IS No D Indicate by check mark whether the registrant file and the submitted of the file of	For the fiscal year ende	ed December 31, 2013
SM ENERGY COMPANY (Exact name of registrant as specified in its charter) Delaware (State or other purisdiction of incorporation or organization) 1775 Sherman Street, Sulter 1200, Denver, Colorado (Address of principal executive offices) Registrant's telephone number, including area code) Securities registered pursuant to Section 12(b) of the Act. This of each class Common stock, 501 par value Securities registered pursuant to Section 12(b) of the Act. None Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes So Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes So No Indicate by check mark whether the registrant (1) has filed all reports required to sele for preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes So No Indicate by check mark whether the registrant (1) has filed all reports required to miss component to Section 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes So No Indicate by check mark whether the registrant (1) has filed all reports required to such site of the such period to such files of the	or	
Delaware (State or other jurisdiction of incorporation or organization)	☐ Transition Report Pursuant to Section 13 or 15(d) of t	the Securities Exchange Act of 1934
(Exact name of registrant as specified in its charter) 1.14.15184.30 (R.R.S. Employer Identification No.) 1.775 Sherman Street, Suite 1200, Denver, Colorado (Address of principal executive offices) (Zip Code) (Zi	Commission file nu	umber 001-31539
(Rate or other jurisdiction of incorporation or organization) 1775 Sherman Street, Sulte 1200, Denver, Colorado (Address of principal executive offices) (Registrant's telephone number, including area code) Securities registered pursuant to Section 12(b) of the Act: Title of each class Securities registered pursuant to Section 12(b) of the Act: New York Stock Exchange Securities registered pursuant to Section 12(g) of the Act: None Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes No \Bar None in the registrant of the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No \Bar None in the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No \Bar None in the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No \Bar None in the registrant is not required to the securities required to the securities for the past 90 days. Yes \Bar No \Bar None in the registrant is not required to the securities to section 13 or Section 15(d) of the Act. Yes No \Bar None in the past 90 days. Yes \Bar No \Bar None in the registrant is a section 15 (d) of the Securities Exchange Act of 1934 during the preceding 2 months (or for such shorter period that the registrant was required to the section 12 (e) to the Securities Exchange Act of 1934 during the preceding 1 not to the section 15 (e) to the Securities Past 9 days. Yes \Bar No \Bar None in the Past 9 days. Yes \Bar None in the Past 9 days of this chapter) is not contained herein, and will not be contained to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this form 10-K or any amendment		
(Address of principal executive offices) (Registrant's telephone number, including area code) Securities registered pursuant to Section 12(b) of the Act: Title of each class Name of each occhange on which registered New York Stock Exchange Securities registered pursuant to Section 12(g) of the Act: None Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes No Indicate by check mark whether the registrant (1) has filed all reports pursuant to Section 13 or Section 15(d) of the Act. Ves No II Indicate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes III Indicate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes III Indicate by check mark whether the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes III Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes III No.	(State or other jurisdiction	
Registrant's telephone number, including area code Securities registered pursuant to Section 12(b) of the Act: Name of each exchange on which registered New York Stock Exchange New York Stock Exchange Securities registered pursuant to Section 12(g) of the Act: None Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes No Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes No Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes No Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes No Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, an accelerated filer, an an accelerated filer, an an accelerated filer, an accelerated filer Smaller reporting company Rule 12b-2 of the Exchange Act. Large accelerated filer Sma		
Name of each exchange on which registered New York Stock Exchange	· ·	
Securities registered pursuant to Section 12(g) of the Act: None Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes No Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Act. Yes No Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and post such files). Yes No Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10 K. Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer Smaller reporting company Smaller reporting company Mon-accelerated filer Smaller reporting compan	Securities registered pursuant	to Section 12(b) of the Act:
Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes⊠ No □ Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes□ No 図 Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 図 No □ Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and posts uch files). Yes 図 No □ Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10 K. 図 Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer □ Smaller reporting company) Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes □ No 図 The aggregate market value of the 65,508,073 shares of voting stock held by non-affiliates of the registrant, based upon the closing sale price of the registrant's common stoon June 28, 2013, the last business day of the registrant's mos		
Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes\box No \Box Research No \Box Registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes\box No \Box Research No \B	, · · · 1	•
Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No Implicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes Implicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes Implicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10 k. Implicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer Accelerated	v i	
Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232,405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes IN Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229,405 of this chapter) is not contained herein, and will not be contained to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10 K. Image accelerated filer, an accelerated filer, an accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer Image accelerated filer, an accelerated filer, an accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer is a smaller reporting company" in Rule 12b-2 of the Exchange Act. Accelerated filer Image accelerated filer in Rule 12b-2 of the Act). Yes Image accelerated filer in Rule 12b-2 of the Act) and Image accelerated filer in Rule 12b-2 of the Act). Yes Image accelerated filer in Rule 12b-2 of the Act) and Image accelerated filer in Rule 12b-2 of the Act). Yes Image accelerated filer in Rule 12b-2 of the Act) and Image accelerated filer in Rule 12b-2 of the Act). Yes Image accelerated filer in Rule 12b-2 of the Act) and Image accelerated filer in Rule 12b-2 of the Act).		
and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes ☑ No ☐ Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10 K. ☑ Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer ☑ Accelerated filer □ Smaller reporting company □ [Do not check if a smaller reporting company] Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes □ No ☑ The aggregate market value of the 65,508,073 shares of voting stock held by non-affiliates of the registrant, based upon the closing sale price of the registrant's common sto on June 28, 2013, the last business day of the registrant's most recently completed second fiscal quarter, of \$59.98 per share, as reported on the New York Stock Exchange; was \$3,929,174,219. Shares of common stock held by each director and executive officer and by each person who owns 10 percent or more of the outstanding common stoc or who is otherwise believed by the registrant to be in a control position have been excluded. This determination of affiliate status is not necessarily a conclusive determination for other purposes. As of February 12, 2014, the registrant had 67,056,441 shares of common stock outstanding, which is net of 22,412 treasury shares held by the registrant. DOCUMENTS INCORPORATED BY REFERENCE	Indicate by check mark whether the registrant (1) has filed all reports required to be filed 12 months (or for such shorter period that the registrant was required to file such reports)	I by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding
to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10 K. Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer Non-accelerated filer Non-accelerated filer Smaller reporting company Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes No The aggregate market value of the 65,508,073 shares of voting stock held by non-affiliates of the registrant, based upon the closing sale price of the registrant's common sto on June 28, 2013, the last business day of the registrant's most recently completed second fiscal quarter, of \$59.98 per share, as reported on the New York Stock Exchange, was \$3,929,174,219. Shares of common stock held by each director and executive officer and by each person who owns 10 percent or more of the outstanding common stoc or who is otherwise believed by the registrant to be in a control position have been excluded. This determination of affiliate status is not necessarily a conclusive determination for other purposes. As of February 12, 2014, the registrant had 67,056,441 shares of common stock outstanding, which is net of 22,412 treasury shares held by the registrant. DOCUMENTS INCORPORATED BY REFERENCE Certain information required by Items 10, 11, 12, 13, and 14 of Part III is incorporated by reference from portions of the registrant's definitive proxy statement relating to	and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the	
"large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. Large accelerated filer Non-accelerated filer Smaller reporting company (Do not check if a smaller reporting company) Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes No The aggregate market value of the 65,508,073 shares of voting stock held by non-affiliates of the registrant, based upon the closing sale price of the registrant's common sto on June 28, 2013, the last business day of the registrant's most recently completed second fiscal quarter, of \$59.98 per share, as reported on the New York Stock Exchange; was \$3,929,174,219. Shares of common stock held by each director and executive officer and by each person who owns 10 percent or more of the outstanding common stoc or who is otherwise believed by the registrant to be in a control position have been excluded. This determination of affiliate status is not necessarily a conclusive determination for other purposes. As of February 12, 2014, the registrant had 67,056,441 shares of common stock outstanding, which is net of 22,412 treasury shares held by the registrant. DOCUMENTS INCORPORATED BY REFERENCE Certain information required by Items 10, 11, 12, 13, and 14 of Part III is incorporated by reference from portions of the registrant's definitive proxy statement relating to	to the best of registrant's knowledge, in definitive proxy or information statements incorp	
Non-accelerated filer Smaller reporting company Smaller reporting company		
(Do not check if a smaller reporting company) Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes□ No ☒ The aggregate market value of the 65,508,073 shares of voting stock held by non-affiliates of the registrant, based upon the closing sale price of the registrant's common sto on June 28, 2013, the last business day of the registrant's most recently completed second fiscal quarter, of \$59.98 per share, as reported on the New York Stock Exchange; was \$3,929,174,219. Shares of common stock held by each director and executive officer and by each person who owns 10 percent or more of the outstanding common stock or who is otherwise believed by the registrant to be in a control position have been excluded. This determination of affiliate status is not necessarily a conclusive determination for other purposes. As of February 12, 2014, the registrant had 67,056,441 shares of common stock outstanding, which is net of 22,412 treasury shares held by the registrant. **DOCUMENTS INCORPORATED BY REFERENCE** Certain information required by Items 10, 11, 12, 13, and 14 of Part III is incorporated by reference from portions of the registrant's definitive proxy statement relating to	Large accelerated filer ⊠	Accelerated filer □
The aggregate market value of the 65,508,073 shares of voting stock held by non-affiliates of the registrant, based upon the closing sale price of the registrant's common sto on June 28, 2013, the last business day of the registrant's most recently completed second fiscal quarter, of \$59.98 per share, as reported on the New York Stock Exchange; was \$3,929,174,219. Shares of common stock held by each director and executive officer and by each person who owns 10 percent or more of the outstanding common stock or who is otherwise believed by the registrant to be in a control position have been excluded. This determination of affiliate status is not necessarily a conclusive determination for other purposes. As of February 12, 2014, the registrant had 67,056,441 shares of common stock outstanding, which is net of 22,412 treasury shares held by the registrant. DOCUMENTS INCORPORATED BY REFERENCE Certain information required by Items 10, 11, 12, 13, and 14 of Part III is incorporated by reference from portions of the registrant's definitive proxy statement relating to		Smaller reporting company \square
on June 28, 2013, the last business day of the registrant's most recently completed second fiscal quarter, of \$59.98 per share, as reported on the New York Stock Exchange; was \$3,929,174,219. Shares of common stock held by each director and executive officer and by each person who owns 10 percent or more of the outstanding common stock or who is otherwise believed by the registrant to be in a control position have been excluded. This determination of affiliate status is not necessarily a conclusive determination for other purposes. As of February 12, 2014, the registrant had 67,056,441 shares of common stock outstanding, which is net of 22,412 treasury shares held by the registrant. DOCUMENTS INCORPORATED BY REFERENCE Certain information required by Items 10, 11, 12, 13, and 14 of Part III is incorporated by reference from portions of the registrant's definitive proxy statement relating to	Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b	o-2 of the Act). Yes□ No⊠
DOCUMENTS INCORPORATED BY REFERENCE Certain information required by Items 10, 11, 12, 13, and 14 of Part III is incorporated by reference from portions of the registrant's definitive proxy statement relating to	on June 28, 2013, the last business day of the registrant's most recently completed secon was \$3,929,174,219. Shares of common stock held by each director and executive office or who is otherwise believed by the registrant to be in a control position have been excluded the control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position have been excluded by the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to be in a control position to the registrant to the registrant to the registrant to the registrant to the	nd fiscal quarter, of \$59.98 per share, as reported on the New York Stock Exchange; er and by each person who owns 10 percent or more of the outstanding common stock
	•	

EXPLANATORY NOTE:

This Amendment No. 1 on Form 10-K/A (the "Amendment") amends the Annual Report on Form 10-K of SM Energy Company (the "Company", "our," "we" or "SM Energy") for the year ended December 31, 2013, originally filed with the Securities and Exchange Commission ("SEC") on February 19, 2014 (the "Original Form 10-K"). This Amendment is an exhibit only filing in response to comments received from the SEC in connection with its review of a request for confidential treatment of certain portions of Exhibit 2.4 originally filed with the Original Form 10-K. In response to these comments, we have amended the redacted portions of Exhibit 2.4. We have modified Part IV Item 15(b), "Exhibits" in this Amendment to reflect our revised request for confidential treatment for Exhibit 2.4. We are re-filing herewith Exhibit 2.4 omitting certain portions for which we have applied for confidential treatment and including others for which we have withdrawn our request for confidential treatment.

In addition, pursuant to the rules of the Securities and Exchange Commission, Item 15 of the Original Form 10-K has been amended to contain currently dated certifications of the Company's Chief Executive Officer and Chief Financial Officer, as required by Section 302 of the Sarbanes-Oxley Act of 2002. All other information contained in the Original Form 10-K remains unchanged.

This Amendment is limited in scope to the items identified above and should be read in conjunction with the Original Form 10-K. This Amendment does not reflect events that have occurred after the filing of the Original Form 10-K and no revisions are being made to the Company's financial statements pursuant to this Amendment. Other than the filing of the information identified above, this Amendment does not modify or update the disclosure in the Original Form 10-K in any way.

2

ITEM 15. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

10.5†

incorporated herein by reference)

(b) Exhibits. The following exhibits are filed or furnished with or incorporated by reference into this report on Form 10-K/A:

Exhibit Number	Description
2.1	Purchase and Sale Agreement dated June 9, 2011, among SM Energy Company, Statoil Texas Onshore Properties LLC, and Talisman Energy USA Inc. (filed as Exhibit 2.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011 and incorporated herein by reference)
2.2	Acquisition and Development Agreement dated June 29, 2011 between SM Energy Company and Mitsui E&P Texas LP (filed as Exhibit 2.2 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011 and incorporated herein by reference)
2.3	First Amendment to Acquisition and Development Agreement dated October 13, 2011 between SM Energy Company and Mitsui E&P Texas (filed as Exhibit 2.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2011 and incorporated herein by reference)
2.4*††	Purchase and Sale Agreement dated November 4, 2013, among SM Energy Company, EnerVest Energy Institutional Fund XIII-A, L.P., EnerVest Energy Institutional Fund XIII-WIB, L.P., and EnerVest Energy Institutional Fund XIII-WIC, L.P.
3.1	Restated Certificate of Incorporation of SM Energy Company, as amended through June 1, 2010 (filed as Exhibit 3.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2010 and incorporated herein by reference)
3.2	Amended and Restated By-Laws of SM Energy Company amended effective as of January 1, 2013 (filed as Exhibit 3.1 to the registrant's Current Report on Form 8-K filed on January 7, 2013, and incorporated herein by reference)
4.1	Indenture related to the 3.50% Senior Convertible Notes due 2027, dated as of April 4, 2007, between St. Mary Land & Exploration Company and Wells Fargo Bank, National Association, as trustee (including the form of 3.50% Senior Convertible Note due 2027) (filed as Exhibit 4.1 to the registrant's Current Report on Form 8-K filed on April 4, 2007 and incorporated herein by reference)
4.2	Indenture related to the 6.625% Senior Notes due 2019, dated as of February 7, 2011, by and between SM Energy Company, as issuer, and U.S. Bank National Association, as Trustee (filed as Exhibit 4.1 to the registrant's Current Report on Form 8-K filed on February 10, 2011, and incorporated herein by reference)
4.3	Indenture related to the 6.50% Senior Notes due 2021, dated as of November 8, 2011, by and among SM Energy Company, as issuer, and U.S. Bank National Association, as Trustee (filed as Exhibit 4.1 to the registrant's Current Report on Form 8-K filed on November 10, 2011, and incorporated herein by reference)
4.4	Indenture related to the 6.50% Senior Notes due 2023, dated June 29, 2012, between SM Energy Company, as Issuer, and U.S. Bank National Association, as Trustee (filed as Exhibit 4.1 to the registrant's Current Report on Form 8-K filed on July 3, 2012, and incorporated herein by reference)
4.5	Indenture related to the 5.0% Senior Notes due 2024, dated May 20, 2013, by and between SM Energy Company, as issuer, and U.S. Bank National Association, as Trustee (filed as Exhibit 4.1 to the registrant's Current Report on Form 8-K filed on May 20, 2013, and incorporated herein by reference)
4.6	Registration Rights Agreement, dated May 20, 2013, by and among SM Energy Company and Merrill Lynch, Pierce, Fenner & Smith Incorporated, Wells Fargo Securities, LLC, and J.P. Morgan Securities LLC, as representatives of several purchasers (filed as Exhibit 4.2 to the registrant's Current Report on Form 8-K filed on May 20, 2013, and incorporated herein by reference)
	3
10.1†	Stock Option Plan, as Amended on May 22, 2003 (filed as Exhibit 99.1 to the registrant's Registration Statement on Form S-8 (Registration No. 333-106438) and incorporated herein by reference)
10.2†	Incentive Stock Option Plan, as Amended on May 22, 2003 (filed as Exhibit 99.2 to the registrant's Registration Statement on Form S-8 (Registration No. 333-106438) and incorporated herein by reference)
10.3†	Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2001 and incorporated herein by reference)
10.4†	Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.9 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2005 and incorporated herein by reference)

Employment Agreement of A.J. Best dated May 1, 2006 (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on May 4, 2006 and

10.6	
10.0	Supplement and Amendment to Deed of Trust, Mortgage, Line of Credit Mortgage, Assignment, Security Agreement, Fixture Filing and Financing Statement for the benefit of Wachovia Bank, National Association, as Administrative Agent, dated effective as of April 14, 2009 (filed as Exhibit 10.2 to the registrant's Current Report on Form 8-K filed on April 20, 2009, and incorporated herein by reference)
10.7	Deed of Trust to Wachovia Bank, National Association, as Administrative Agent, dated effective as of April 14, 2009 (filed as Exhibit 10.3 to the registrant's Current Report on Form 8-K filed on April 20, 2009, and incorporated herein by reference)
10.8†	Equity Incentive Compensation Plan as Amended and Restated as of March 26, 2009 (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on May 27, 2009, and incorporated herein by reference)
10.9†	Equity Incentive Compensation Plan As Amended and Restated as of April 1, 2010 (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on June 2, 2010, and incorporated herein by reference)
10.10♦	SM Energy Company Equity Incentive Compensation Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.7 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference)
10.11†	Third Amendment to Employee Stock Purchase Plan dated September 23, 2009 (filed as Exhibit 10.3 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2009, and incorporated herein by reference)
10.12†	Fourth Amendment to Employee Stock Purchase Plan dated December 29, 2009 (filed as Exhibit 10.46 to the registrant's Annual Report on Form 10-K for the year ended December 31, 2009, and incorporated herein by reference)
10.13♦	Employee Stock Purchase Plan, As Amended and Restated as of July 30, 2010 (filed as Exhibit 10.4 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference)
10.14†	Form of Performance Share and Restricted Stock Unit Award Agreement as of July 1, 2010 (filed as Exhibit 10.3 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2010 and incorporated herein by reference)
10.15†	Form of Performance Share and Restricted Stock Unit Award Notice as of July 1, 2010 (filed as Exhibit 10.4 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2010 and incorporated herein by reference)
10.16†	Form of Non-Employee Director Restricted Stock Award Agreement as of May 27, 2010 (filed as Exhibit 10.5 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2010 and incorporated herein by reference)
10.17**	Gas Services Agreement effective as of July 1, 2010 between SM Energy Company and Eagle Ford Gathering LLC (filed as Exhibit 10.3 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference)
10.18♦	Cash Bonus Plan, As Amended on July 30, 2010 (filed as Exhibit 10.5 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference)
	4
10.19♦	Net Profits Interest Bonus Plan, As Amended by the Board of Directors on July 30, 2010 (filed as Exhibit 10.6 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference)
10.19 ♦ 10.20 ♦	
	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the
10.20♦	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-
10.20 ♦ 10.21†	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K
10.20♦ 10.21† 10.22†	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Pension Plan for Employees of SM Energy Company as Amended and Restated as of January 1, 2010 (filed as Exhibit 10.30 to the registrant's Annual
10.20♦ 10.21† 10.22† 10.23	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Pension Plan for Employees of SM Energy Company as Amended and Restated as of January 1, 2010 (filed as Exhibit 10.30 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan as Amended as of November 9, 2010 (filed as Exhibit 10.31 to the
10.20♦ 10.21† 10.22† 10.23 10.24+	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Pension Plan for Employees of SM Energy Company as Amended and Restated as of January 1, 2010 (filed as Exhibit 10.30 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan as Amended as of November 9, 2010 (filed as Exhibit 10.31 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Fourth Amended and Restated Credit Agreement dated May 27, 2011 among SM Energy Company, Wells Fargo Bank, National Association, as Administrative Agent, and the Lenders party thereto (filed as Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June
10.20♦ 10.21† 10.22† 10.23 10.24+ 10.25	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Pension Plan for Employees of SM Energy Company as Amended and Restated as of January 1, 2010 (filed as Exhibit 10.30 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan as Amended as of November 9, 2010 (filed as Exhibit 10.31 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Fourth Amended and Restated Credit Agreement dated May 27, 2011 among SM Energy Company, Wells Fargo Bank, National Association, as Administrative Agent, and the Lenders party thereto (filed as Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference) Gas Gathering Agreement dated May 31, 2011 between Regency Field Services LLC and SM Energy Company (filed as Exhibit 10.2 to the registrant's
10.20♦ 10.21† 10.22† 10.23 10.24+ 10.25	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Pension Plan for Employees of SM Energy Company as Amended and Restated as of January 1, 2010 (filed as Exhibit 10.30 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan as Amended as of November 9, 2010 (filed as Exhibit 10.31 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Fourth Amended and Restated Credit Agreement dated May 27, 2011 among SM Energy Company, Wells Fargo Bank, National Association, as Administrative Agent, and the Lenders party thereto (filed as Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference) Gas Gathering Agreement dated May 31, 2011 between Regency Field Services LLC and SM Energy Company (filed as Exhibit 10.2 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference)
10.20♦ 10.21† 10.22† 10.23 10.24+ 10.25 10.26	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Pension Plan for Employees of SM Energy Company as Amended and Restated as of January 1, 2010 (filed as Exhibit 10.30 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan as Amended as of November 9, 2010 (filed as Exhibit 10.31 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Fourth Amended and Restated Credit Agreement dated May 27, 2011 among SM Energy Company, Wells Fargo Bank, National Association, as Administrative Agent, and the Lenders party thereto (filed as Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference) Gas Gathering Agreement dated May 31, 2011 between Regency Field Services LLC and SM Energy Company (filed as Exhibit 10.2 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference) Gasthering and Natural Gas Services Agreement effective as of April 1, 2011 between SM Energy Company and ETC Texas Pipeline, Ltd. (filed as Exhibit 10.3 to the registrant's Qua
10.20♦ 10.21† 10.22† 10.23 10.24+ 10.25 10.26 10.27	Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan, As Amended as of July 30, 2010 (filed as Exhibit 10.8 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2010 and incorporated herein by reference) Form of Amendment to Form of Change of Control Executive Severance Agreement (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on December 29, 2010, and incorporated herein by reference) Amendment to A.J. Best Employment Agreement dated December 31, 2010 (filed as Exhibit 10.28 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Pension Plan for Employees of SM Energy Company as Amended and Restated as of January 1, 2010 (filed as Exhibit 10.30 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) SM Energy Company Non-Qualified Unfunded Supplemental Retirement Plan as Amended as of November 9, 2010 (filed as Exhibit 10.31 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2010, and incorporated herein by reference) Fourth Amended and Restated Credit Agreement dated May 27, 2011 among SM Energy Company, Wells Fargo Bank, National Association, as Administrative Agent, and the Lenders party thereto (filed as Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference) Gas Gathering Agreement dated May 31, 2011 between Regency Field Services LLC and SM Energy Company (filed as Exhibit 10.2 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference) Gas Processing Agreement effective as of April 1, 2011 between ETC Texas Pipeline, Ltd. and SM Energy Company (filed as Exhibit 10.4 to the registrant's Quarterly Report on Form

Form of Restricted Stock Unit Award Agreement as of July 1, 2011 (filed as Exhibit 10.7 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, and incorporated herein by reference)

10.31†

10.32†	Form of Performance Stock Unit Award Agreement as of September 6, 2011 (filed as Exhibit 10.1 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2011, and incorporated herein by reference)		
10.33†	Form of Restricted Stock Unit Award Agreement as of September 6, 2011 (filed as Exhibit 10.2 to the registrant's Quarterly Report on Form 10-Q for the quarter ended September 30, 2011, and incorporated herein by reference)		
10.34	Amendment No. 1 to the Pension Plan for Employees of SM Energy Company amended as of January 1, 2011 (filed as Exhibit 10.41 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2011, and incorporated herein by reference)		
10.35	Amendment No. 2 to the Pension Plan for Employees of SM Energy Company amended as of January 1, 2012 (filed as Exhibit 10.42 to the registrant's Annual Report on Form 10-K filed for the year ended December 31, 2011, and incorporated herein by reference)		
10.36†	Equity Incentive Compensation Plan, As Amended as of May 22, 2013 (filed as Annex A to the registrant's Schedule 14A filed on April 11, 2013, and incorporated herein by reference)		
	5		
10.37	Fifth Amended and Restated Credit Agreement dated April 12, 2013, among SM Energy Company, as Borrower, Wells Fargo Bank, National Association, as Administrative Agent, and the Lenders party thereto (filed as Exhibit 10.1 to the registrant's Current Report of Form 8-K filed on April 15, 2013, and incorporated herein by reference)		
10.38†	Form of Performance Stock Unit Award Agreement as of July 31, 2013 (filed as Exhibit 10.2 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2013, and incorporated herein by reference)		
10.39†	Form of Restricted Stock Unit Award Agreement as of July 31, 2013 (filed as Exhibit 10.3 to the registrant's Quarterly Report on Form 10-Q for the quarter ended June 30, 2013, and incorporated herein by reference)		
10.40†	SM Energy Company Non-Qualified Deferred Compensation Plan as of March 10, 2014 (filed as Exhibit 10.1 to the registrant's Current Report on Form 8-K filed on January 24, 2014, and incorporated herein by reference)		
10.41†++	Cash Bonus Plan, As Amended and Restated on February 1, 2014		
10.42†++	Summary of Compensation Arrangements for Non-Employee Directors		
12.1++	Computation of Ratio of Earnings to Fixed Charges		
21.1++	Subsidiaries of Registrant		
23.1++	Consent of Ernst & Young LLP		
23.2++	Consent of Deloitte & Touche LLP		
23.3++	Consent of Ryder Scott Company L.P.		
24.1++	Power of Attorney		
31.1*	Certification of Chief Executive Officer pursuant to Section 302 of the Sarbanes — Oxley Act of 2002		
31.2*	Certification of Chief Financial Officer pursuant to Section 302 of the Sarbanes — Oxley Act of 2002		
32.1++	Certification pursuant to 18 U.S.C. Section 1350 as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002		
99.1++	Ryder Scott Audit Letter		
101.INS++	XBRL Instance Document		
101.SCH++	XBRL Schema Document		
101.CAL++	XBRL Calculation Linkbase Document		
101.LAB++	XBRL Label Linkbase Document		
101.PRE++	XBRL Presentation Linkbase Document		
101.DEF++	XBRL Taxonomy Extension Definition Linkbase Document		
* 1:	lad with this Forms 10 V/A		

Filed with this Form 10-K/A.

Certain portions of this exhibit have been redacted and are subject to a confidential treatment order granted by the Securities and Exchange Commission pursuant to Rule 24b-2 under the Securities Exchange Act of 1934.

Exhibit constitutes a management contract or compensatory plan or agreement.

Confidential Treatment has been requested with respect to portions of the exhibit. Such portions have been redacted and filed separately with the SEC. Exhibit constitutes a management contract or compensatory plan or agreement. This document was amended on July 30, 2010 primarily to reflect the recent change in the name of the registrant from St. Mary Land & Exploration Company to SM Energy Company. There were no material changes to the substantive terms and conditions in this document.

Exhibit constitutes a management contract or compensatory plan or agreement. This document was amended on November 9, 2010, in order to make technical revisions to ensure compliance with Section 409A of the Internal Revenue Code. There were no material changes to the substantive terms and conditions in this document.

Exhibit was previously filed or furnished, as applicable, with the Company's original Annual Report on Form 10-K for the year ended December 31, 2013 which was originally filed on February 19, 2014 and is incorporated herein by reference.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

SM ENERGY COMPANY

(Registrant)

Date: May 9, 2014

By: /s/ ANTHONY J. BEST

Anthony J. Best Chief Executive Officer (Principal Executive Officer)

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

Signature	Title	Date
/s/ ANTHONY J. BEST Anthony J. Best	Chief Executive Officer and Director (Principal Executive Officer)	May 9, 2014
/s/ A. WADE PURSELL A. Wade Pursell	Executive Vice President and Chief Financial Officer (Principal Financial Officer)	May 9, 2014
/s/ MARK T. SOLOMON Mark T. Solomon	Vice President - Controller and Assistant Secretary (Principal Accounting Officer)	May 9, 2014

7

Signature	Title	Date
Vs/ * William D. Sullivan	Chairman of the Board of Directors	May 9, 2014
/s/ * Barbara M. Baumann	Director	May 9, 2014
/s/ * Larry W. Bickle	Director	May 9, 2014
/s/ * Stephen R. Brand	Director	May 9, 2014
/s/ * William J. Gardiner	Director	May 9, 2014
/s/ * Loren M. Leiker	Director	May 9, 2014
/s/ * Julio M. Quintana	Director	May 9, 2014
/s/ * John M. Seidl	Director	May 9, 2014
* /s/ A. Wade Pursell By A. Wade Pursell Attorney in Fact		

8

CONFIDENTIAL TREATMENT REQUESTED BY SM ENERGY COMPANY

Portions of this Exhibit have been redacted pursuant to a request for confidential treatment under Rule 24b-2 of the General Rules and Regulations under the Securities

Exchange Act. Omitted information marked "[***]" in this Exhibit has been filed with the Securities and Exchange Commission together with such request for confidential treatment.

Execution Copy

PURCHASE AND SALE AGREEMENT

BETWEEN

SM ENERGY COMPANY

as Seller

and

ENERVEST ENERGY INSTITUTIONAL FUND XIII-A, L.P.,

ENERVEST ENERGY INSTITUTIONAL FUND XIII-WIB, L.P., AND

ENERVEST ENERGY INSTITUTIONAL FUND XIII-WIC, L.P.

collectively, as Buyer

DATED November 4, 2013

TABLE OF CONTENTS

		Page
Article I DEFINITIO	NS; INTERPRETATION	1
1.1	Definitions	1
1.2	Interpretation	1
Article II Purchase ar	nd Sale	2
2.1	Purchase and Sale	2
2.2	Purchase Price	2
2.3	Deposit	2
2.4	Adjustments to Purchase Price	3
2.5	Allocated Value	5
2.6	Settlement; Disputes	5
2.7	Section 1031 Like-Kind Exchange	6
Article III REPRESE	NTATIONS AND WARRANTIES OF SM ENERGY	7
3.1	Organization, Existence	7
3.2	Authorization	7
3.3	No Conflicts	7
3.4	Bankruptcy	7
3.5	Foreign Person	7
3.6	Litigation	8
3.7	Material Contracts	8
3.8	No Violation of Laws	9
3.9	Royalties, Etc.	9
3.10	Imbalances	9
3.11	Current Commitments	9
3.12	Environmental	9
3.13	Asset Taxes	9
3.14	Brokers' Fees	9
3.15	Advance Payments	9
3.16	Partnerships	10
3.17	Payout Balances	10
3.18	Preferential Rights to Purchase and Consents	10
3.19	No Other Representations or Warranties; Disclosed Materials	10
3.19	No Other Representations of Warrannes, Disclosed Materials	10
Article IV REPRESE	NTATIONS AND WARRANTIES OF BUYER	10
4.1	Organization; Existence	10
4.2	Authorization	11
4.3	No Conflicts	11
4.4	Consents	11
4.5	Bankruptcy	11

4.6	Litigation	11
4.7	Financing	11
4.8 4.9	Independent Evaluation Brokers' Fees	12 12
4.10	Accredited Investor	12
Antiala V ACCESS	/ DICCLAIMEDC	12
Article V ACCESS	DISCLAIMERS	12
5.1	Access	12
5.2 5.3	Confidentiality Disclaimers	14 14
5.5	Discianners	14
Article VI TITLE M	ATTERS; CASUALTIES	16
6.1	SM Energy's Title	16
6.2	Notice of Title Defects; Defect Adjustments	16
6.3	Casualty or Condemnation Loss	20
6.4	Preferential Rights and Consents to Assign	21
Article VII ENVIRO	DNMENTAL MATTERS	23
7.1	Farrian and A. D. forte	22
7.1 7.2	Environmental Defects NORM, Wastes and Other Substances	23 25
Article VIII CERTA	IN AGREEMENTS	25
8.1	Conduct of Business	25
8.2	Governmental Bonds	27
8.3	Notifications	27
8.4 8.5	Suspense Accounts Digital Records	27 27
8.6	Chieftain Litigation	28
A -4:-1- IV CONDIT	IONG TO CLOSING	20
Article IX CONDIT	IONS TO CLOSING	28
9.1	Buyer's Conditions to Closing	28
9.2	SM Energy's Conditions to Closing	29
Article X Tax Matte	rs	29
10.1	Asset Tax Liability	29
10.2 10.3	Transfer Taxes Asset Tax Reports and Returns	30 30
10.4	Tax Cooperation	30
Ati-1- VI CI OCINI	9	21
Article XI CLOSIN	J	31
11.1	Date of Closing	31
		31
		_
11.2	Place of Closing	31
11.3	Closing Obligations	31
11.4	Records	32
Article XII ACQUIS	SITION TERMINATION AND REMEDIES	32
12.1 12.2	Right of Termination Effect of Termination	32 32
12.3	Specific Performance	33
12.4	Return of Documentation and Confidentiality	33
Article XIII ASSUM	MPTION; SURVIVAL; INDEMNIFICATION	33
13.1	Assumption by Buyer	33
13.2	Indemnities of SM Energy	34
13.3	Indemnities of Buyer	34
13.4	Limitation on Liability	35
13.5 13.6	Express Negligence Exclusive Remedy for Agreement	35 35
13.7	Indemnification Procedures	36
13.8	Survival	38
13.9	Non-Compensatory Damages	38
Article XIV MISCE	LLANEOUS	39
14.1	Counterparts	39
14.2	Notices	39
14.3 14.4	Expenses Waivers; Rights Cumulative	40 40
17.7	martin, ragina Camalanye	40

14.5	Relationship of the Parties	41
14.6	Entire Agreement; Conflicts	41
14.7	Governing Law	41
14.8	Filings, Notices and Certain Governmental Approvals	42
14.9	Amendment	42
14.10	Parties in Interest	42
14.11	Successors and Permitted Assigns	42
14.12	Publicity	42
14.13	Preparation of Agreement	42
14.14	Severability	42

Annexes

Annex I Definitions

Schedules

Schedule 2.5	Allocated Values
Schedule 3.6	Litigation
Schedule 3.7(a)	Material Contracts
Schedule 3.8	Violation of Laws
Schedule 3.9	Royalties
Schedule 3.10	Imbalances
Schedule 3.11	AFEs
Schedule 3.12	Environmental Matters
Schedule 3.13	Asset Taxes
Schedule 3.16	Partnerships
Schedule 3.17	Payout Balances
Schedule 6.4	Consents and Preferential Rights
Schedule 8.1	Permitted Operations

Exhibits

Exhibit A	Leases
Exhibit B	Wells and Well Locations
Exhibit C	Form of Assignment
Exhibit D-1	Braithwaite Area
Exhibit D-2	Broxton Area
Exhibit D-3	Cleveland/Tonkawa Area
Exhibit D-4	Mayfield/Sixty Six Area
Exhibit D-5	Woodford Area
Exhibit E	Buyer's Proportionate Shares

Exhibit F Excluded Assets

THIS PURCHASE AND SALE AGREEMENT (this "Agreement") is made as of November 4, 2013 (the "Execution Date") between SM ENERGY COMPANY, a Delaware corporation ("SM Energy") and ENERVEST ENERGY INSTITUTIONAL FUND XIII-A, L.P., ENERVEST ENERGY INSTITUTIONAL FUND XIII-WIB, L.P., and ENERVEST ENERGY INSTITUTIONAL FUND XIII-WIC, L.P., each a Delaware limited partnership, (collectively, "Buyer"). SM Energy and Buyer shall sometimes be referred to herein together as the "Parties", and each individually as a "Party".

RECITALS

WHEREAS, SM Energy owns certain oil and gas leases located in Texas and Oklahoma and associated assets as more fully described below; and

WHEREAS, SM Energy desires to sell and Buyer desires to purchase, in their respective undivided proportionate interests set forth opposite each Buyer's name in **Exhibit E** hereto (each Buyer's "**Proportionate Share**"), the Assets (as defined below) upon the terms and conditions set forth in this Agreement.

NOW THEREFORE, in consideration of the mutual agreements herein contained, the Parties hereby agree as follows:

ARTICLE I DEFINITIONS; INTERPRETATION

- 1.1 Definitions. Capitalized terms used in this Agreement shall have the meanings given such terms in Annex I.
- 1.2 Interpretation. All references in this Agreement to Exhibits, Appendices, Annexes, Articles, Sections, subsections and other subdivisions of or to this Agreement unless expressly provided otherwise. Titles appearing at the beginning of any Articles, Sections, subsections and other subdivisions of this Agreement are for convenience only, do not constitute any part of this Agreement, and shall be disregarded in construing the language hereof. The words "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular Article, Section, subsection or other subdivision unless expressly so limited. The words "this Article," "this Section" and "this subsection," and words of similar import, refer only to Article, Section or subsection hereof in which such words occur. The word "including" (in its various forms) means "including without limitation." All references to "\$" or "dollars" shall be deemed references to United States Dollars. Each accounting term not defined herein will have the meaning given to it under GAAP, as in effect on the Execution Date. Pronouns in masculine, feminine or neuter genders shall be construed to state and include any other gender, and words, terms and titles (including terms defined herein) in the singular form shall be construed to include the plural and vice versa, unless the context otherwise requires. Appendices, Annexes and Exhibits referred to herein are attached to and made a part of this Agreement. Unless expressly stated otherwise, references to any Law or contract shall mean such Law or contract as it may be amended from time to time.

ARTICLE II PURCHASE AND SALE

2.1 Purchase and Sale.

- (a) Subject to the terms and conditions of this Agreement, Buyer agrees to purchase from SM Energy, and SM Energy agrees to sell, assign and deliver to Buyer the Assets for the consideration specified in this *Article II*.
 - (b) SM Energy shall reserve and retain all of the Excluded Assets.
- **2.2** *Purchase Price*. Subject to the other terms and conditions of this Agreement, the purchase price for the Assets shall be Three Hundred Forty-Three Million Dollars (\$343,000,000.00) (the "*Purchase Price*"). The Purchase Price shall be payable as follows:
- (a) Not later than three (3) Business Days after the execution and delivery of this Agreement, Buyer will deliver or cause to be delivered to an account (the "Escrow Account") with Wells Fargo Bank, National Association (the "Escrow Agent") a wire transfer in the amount equal to 5% of unadjusted Purchase Price (such amount, together with any interest or other income earned thereon, the "Deposit") to be held, invested and disbursed in accordance with the terms of this Agreement and an escrow agreement among SM Energy, Buyer and Escrow Agent (the "Escrow Agreement"). The Deposit shall be distributed to SM Energy in accordance with Section 11.3(e) if the Closing occurs or shall be otherwise distributed in accordance with the terms of Section 2.3.
 - (b) At Closing, Buyer shall pay SM Energy the Closing Amount by wire transfer in immediately available funds.

2.3 Deposit.

- (a) If the transactions contemplated by Section 2.1 are not consummated on or before the Outside Termination Date because of: (i) the failure of Buyer to materially perform any of its obligations hereunder, or (ii) the failure of any of Buyer's representations or warranties hereunder to be true and correct in all material respects as of the date of this Agreement and as of the Closing, then, in such event, SM Energy shall have the right to terminate this Agreement and if SM Energy elects to exercise that termination right, then SM Energy and Buyer shall instruct the Escrow Agent to distribute the Deposit to SM Energy free of any claims by Buyer thereto. In such event, the Deposit shall be distributed to SM Energy as liquidated damages, which remedy shall be the sole and exclusive remedy available to SM Energy, and SM Energy expressly waives all other remedies, legal or equitable that it may have. Buyer and SM Energy acknowledge and agree that (i) SM Energy's actual damages upon the event of such a termination are difficult to ascertain with any certainty, (ii) the Deposit is a reasonable estimate of such actual damages, and (iii) such liquidated damages do not constitute a penalty.
- (b) If this Agreement is terminated by the mutual written agreement of the Parties, or if the Closing does not occur on or before the Outside Termination Date for any reason other than as set forth in *Section 2.3(a)*, then Buyer shall be entitled to the delivery of the

2

Deposit, and SM Energy and Buyer shall instruct the Escrow Agent to distribute the Deposit to Buyer free of any claims by SM Energy with respect thereto.

(c) In the event of any termination of this Agreement as contemplated by Section 2.3(a) or Section 2.3(b) above, the Parties shall, in each case, thereupon have the rights and obligations set forth in Section 12.2 and Section 12.3.

2.4 Adjustments to Purchase Price.

- (a) For purposes of determining the amounts of the adjustments to the Purchase Price provided for in this Section 2.4, the principles set forth in this Section 2.4(a) shall apply. Buyer shall be entitled to all production of Hydrocarbons from or attributable to the Units, Leases, and Wells at and after the Effective Time (and all products and proceeds attributable thereto), and to all other income, proceeds, receipts and credits earned with respect to the Assets at or after the Effective Time, and shall be responsible for (and entitled to any refunds with respect to) all Operating Expenses incurred at and after the Effective Time. SM Energy shall be entitled to all Hydrocarbon production from or attributable to Units, Leases and Wells prior to the Effective Time (and all products and proceeds attributable thereto), and to all other income, proceeds, receipts and credits earned with respect to the Assets prior to the Effective Time, and shall be responsible for (and entitled to any refunds with respect to) all Operating Expenses incurred prior to the Effective Time. "Earned" and "incurred", as used in the Agreement shall be interpreted in accordance with GAAP and Council of Petroleum Accountants Society standards, except as otherwise specified herein. For purposes of allocating production (and proceeds and accounts receivable with respect thereto), under this Section 2.4, (i) liquid Hydrocarbons shall be deemed to be "from or attributable to" the Units, Leases and Wells when they pass through the royalty measurement meters, delivery point sales meters or custody transfer meters on the gathering lines or pipelines through which they are transported (whichever meter is closest to the well). SM Energy shall utilize reasonable interpolative procedures, consistent with industry practice, to arrive at an allocation of production when exact meter readings or gauging and strapping data are not available. As part of the Preliminary Settlement Statement, Buyer shall provide to SM Energy such data as is reasonably necessar
 - (b) The Purchase Price shall be adjusted upward by the following amounts (without duplication):
 - (i) an amount equal to the value of all Hydrocarbons produced from or attributable to the Assets in storage or existing in stock tanks, pipelines and/or plants (including inventory) as of the Effective Time, the value to be based upon the contract price in effect as of the Effective Time (or if there is no contract price, then the market price in effect as of the Effective Time in the field in which such Hydrocarbons were produced or if actually sold prior to the date of determination, the proceeds actually recovered by SM Energy attributable to such sale), net of amounts payable as royalties, overriding royalties and other burdens upon, measured by or payable out of such production and Asset Taxes;

3

- (ii) an amount equal to all Operating Expenses (other than Asset Taxes) attributable to the Assets from and after the Effective Time that were paid by SM Energy or its Affiliates;
- (iii) to the extent that SM Energy is under-produced or has over-delivered at the wellhead or into pipelines (in each case) as of the Effective Time as shown with respect to the net Imbalances set forth in *Schedule 3.10*, as complete and final settlement of all such Imbalances, to the extent attributable to the Assets and subject to *Section 2.4(d)*, the sum of \$1,408,535 (based on a value of \$1.50 per Mcf);
 - (iv) Subject Transfer Taxes paid by SM Energy with respect to the transactions contemplated by this Agreement; and

- (v) any other amount provided for elsewhere in this Agreement or otherwise agreed upon by the Parties.
- (c) The Purchase Price shall be adjusted downward by the following amounts (without duplication):
- (i) an amount equal to all proceeds actually received by SM Energy attributable to the sale of Hydrocarbons (A) produced from or attributable to the Assets during the Interim Period or (B) contained in storage or existing in stock tanks, pipelines and/or plants (including inventory) as of the Effective Time for which an upward Purchase Price adjustment was made pursuant to Section 2.4(b);
- (ii) if SM Energy makes the election under *Section 6.2(c)(i)* with respect to a Title Defect, the Title Defect Amount with respect to such Title Defect if the Title Defect Amount has been determined prior to Closing;
- (iii) if SM Energy makes the election under Section 7.1(b)(i) with respect to an Environmental Defect, the Remediation Amount with respect to such Environmental Defect if the Remediation Amount has been determined prior to Closing;
- (iv) an amount determined pursuant to Section 7.1(b)(ii), Section 6.4(c)(ii) or Section 6.4(d)(i), as applicable, for any Asset excluded from the transaction contemplated hereby pursuant to such Sections;
 - (v) the value of any Casualty Loss pursuant to Section 6.3(b)(B);
- (vi) an amount equal to all Operating Expenses attributable to the Assets prior to the Effective Time that were paid by Buyer or its Affiliates; and
 - (vii) any other amount provided for elsewhere in this Agreement or otherwise agreed upon by the Parties.
- (d) If, prior to Closing, Buyer or SM Energy discovers an error in the Imbalances set forth in Schedule 3.10, then the Purchase Price shall be further adjusted at

4

Closing based on a value of \$1.50 per Mcf, and Schedule 3.10 will be deemed amended immediately prior to the Closing to reflect the Imbalances for which the Purchase Price is so adjusted.

2.5 Allocated Value. SM Energy and Buyer have agreed upon an allocation of the unadjusted Purchase Price among each of the Wells and Well Locations, and such allocations are set forth on Schedule 2.5 (the "Allocated Value"). SM Energy and Buyer agree that the Allocated Values shall be used in calculating adjustments to the Purchase Price as provided herein. Any adjustments to the Purchase Price, other than the adjustments provided for in Section 2.4, shall be applied on a pro rata basis to the amounts set forth in Schedule 2.5, as applicable, for all Assets. After all such adjustments are made, any adjustments to the Purchase Price made pursuant toSection 2.4 shall be applied to the amounts set forth on Schedule 2.5, as applicable, for the particular affected Assets. For tax purposes, the Parties agree to report the transactions contemplated by this Agreement in a manner consistent with the terms of this Agreement, including the allocations set forth above as of the Closing Date, and that neither Party will take any position inconsistent therewith, including in any tax return, refund claim, litigation, arbitration, or otherwise.

2.6 Settlement; Disputes.

- (a) Not less than five (5) Business Days prior to the Closing, SM Energy shall prepare and submit to Buyer for review a draft settlement statement using the best information available to SM Energy (the "*Preliminary Settlement Statement*") that shall set forth the adjusted Purchase Price reflecting each adjustment made in good faith in accordance with this Agreement as of the date of preparation of such Preliminary Settlement Statement. Within three (3) Business Days of receipt of the Preliminary Settlement Statement, Buyer has the right, but not the obligation, to deliver to SM Energy a written report containing all changes with the explanation therefor that Buyer proposes to be made to the Preliminary Settlement Statement. The Preliminary Settlement Statement, as agreed upon by the Parties, will be used to adjust the Purchase Price at Closing (such adjusted price, the "*Preliminary Purchase Price*"). If the Parties cannot agree on the Preliminary Settlement Statement prior to the Closing, the Preliminary Settlement Statement as presented by SM Energy will be used to adjust the Purchase Price at Closing.
- (b) On or before 120 days after the Closing, a final settlement statement (the "Final Settlement Statement") will be prepared by SM Energy, based on actual income and expenses during the Interim Period and which takes into account all final adjustments made to the Purchase Price, including any related adjustments for Title Defect Properties pursuant to Section 6.2, and shows the resulting final adjusted Purchase Price (the "Final Purchase Price"). SM Energy shall, at Buyer's request, supply reasonable documentation in its or its Affiliates possession available to support the actual revenue, expenses and other items for which adjustments are made. The Final Settlement Statement shall set forth the actual proration of the amounts required by this Agreement. As soon as practicable, and in any event within 45 days, after receipt of the Final Settlement Statement, Buyer has the right, but not the obligation, to return a written report containing any proposed changes to the Final Settlement Statement and an explanation of any such changes and the reasons therefor (the "Dispute Notice"). If the Final Purchase Price set forth in the Final Settlement Statement is mutually agreed upon by SM

5

Energy and Buyer, the Final Settlement Statement and the Final Purchase Price shall be final and binding on the Parties. Once the Final Purchase Price has been agreed upon by the Parties pursuant to this *Section 2.6(b)* or determined by the Accounting Arbitrator pursuant to *Section 2.6(c)*, as applicable, the Parties shall execute a certificate setting forth such agreed or determined, as applicable, Final Purchase Price, which shall be binding on the Parties for all purposes of this Agreement. The Parties will promptly make the payment necessary to true-up to the Final Purchase Price.

- (c) If the Parties are unable to resolve the matters addressed in the Dispute Notice, each of Buyer and SM Energy shall, within 14 Business Days following the delivery of such Dispute Notice, summarize its position with regard to such dispute in a written document and submit such summaries to the Houston, Texas office of KPMG LLP or such other Person as the Parties may mutually select (the "Accounting Arbitrator"), together with the Dispute Notice, the Final Settlement Statement and any other documentation such Party may desire to submit. Within 20 Business Days after receiving the Parties' respective submissions, the Accounting Arbitrator shall render a decision choosing either SM Energy's position or Buyer's position with respect to each matter addressed in any Dispute Notice, based on the materials described above. Any decision rendered by the Accounting Arbitrator pursuant hereto shall be final, conclusive and binding on the Parties and will be enforceable against any of the Parties in any court of competent jurisdiction. The costs of such Accounting Arbitrator shall be borne one-half by Buyer and one-half by SM Energy.
- 2.7 Section 1031 Like-Kind Exchange. SM Energy and Buyer hereby agree that SM Energy shall have the right at any time prior to completion of all the transactions that are to occur at Closing to assign all or a portion of its rights under this Agreement to a Qualified Intermediary (as that term is defined in Section 1.1031(k)-1(g)(4)(v) of the Treasury Regulations) in order to accomplish the transaction in a manner that will comply, either in whole or in part, with the requirements of a like-kind exchange pursuant to Section 1031 of the Code. Likewise, Buyer shall have the right at any time prior to completion of all the transactions that are to occur at Closing to assign all or a portion of its rights under this Agreement to a Qualified Intermediary for the same purpose. If SM Energy assigns all or any of its rights under this Agreement

for this purpose, Buyer agrees to (a) consent to SM Energy's assignment of its rights in this Agreement, which assignment shall be in a form reasonably acceptable to Buyer, and (b) pay the Purchase Price (or a designated portion thereof as specified by SM Energy) into a qualified escrow or qualified trust account(s) at Closing as directed in writing. If Buyer assigns all or any of its rights under this Agreement for this purpose, SM Energy agrees to (i) consent to Buyer's assignment of its rights in this Agreement, which assignment shall be in a form reasonably acceptable to SM Energy, (ii) accept the Purchase Price from the qualified escrow or qualified trust account at Closing, and (iii) at Closing, convey and assign directly to Buyer the Assets (or any portion thereof) as directed by Buyer. SM Energy and Buyer acknowledge and agree that any assignment of this Agreement (or any rights hereunder) to a Qualified Intermediary shall not release any Party from any of its respective liabilities and obligations hereunder, and that neither Party represents to the other Party that any particular tax treatment will be given to any Party as a result thereof. The Party electing to assign all or any of its rights under this Agreement pursuant to this Section 2.7 shall defend, indemnify, and hold harmless the other Party and its Affiliates from all claims relating to such election.

6

ARTICLE III REPRESENTATIONS AND WARRANTIES OF SM ENERGY

SM Energy represents and warrants to Buyer as follows:

- 3.1 Organization, Existence. SM Energy is a corporation duly formed and validly existing under the Laws of the State of Delaware. SM Energy has all requisite power and authority to own and operate its property (including, without limitation, the Assets) and to carry on its business as now conducted. SM Energy is duly licensed or qualified to do business as a foreign corporation and is in good standing in all jurisdictions in which such qualification is required by Law, except where the failure to qualify or be in good standing would not have a Material Adverse Effect.
- 3.2 Authorization. SM Energy has full power and authority to enter into and perform this Agreement and the transactions contemplated herein. The execution, delivery and performance by SM Energy of this Agreement has been, and the execution, delivery and performance by SM Energy of all other documents delivered pursuant to this Agreement will be when delivered, duly and validly authorized and approved by all necessary corporate action on the part of SM Energy. Assuming the due authorization, execution and delivery by the other parties to such documents, this Agreement constitutes, and the other documents delivered pursuant to this Agreement to which SM Energy is a party will constitute when delivered, SM Energy's legal, valid and binding obligations, enforceable against SM Energy in accordance with their respective terms, subject however to the effects of bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and similar Laws relating to or affecting creditors' rights, as well as to principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at Law).
- 3.3 No Conflicts. Assuming the receipt of all Consents and the waiver of all Preferential Rights, the execution, delivery and performance by SM Energy of this Agreement and the consummation of the transactions contemplated hereby does not and will not (a) conflict with or result in a breach of any provisions of the organizational documents or other governing documents of SM Energy, (b) except for Permitted Encumbrances, result in a default or the creation of any Encumbrance or give rise to any right of termination, cancellation or acceleration under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license or other material agreement (other than the Leases or Applicable Contracts) to which SM Energy is a party or by which SM Energy or the Assets may be bound or (c) violate any Law applicable to SM Energy or any of the Assets, except in the case of clauses (b) and (c) where such default, Encumbrance, termination, cancellation, acceleration or violation would not have a material adverse effect upon the ability of SM Energy to consummate the transactions contemplated by this Agreement.
- 3.4 Bankruptcy. There are no bankruptcy, reorganization or receivership proceedings pending, being contemplated by or, to SM Energy' Knowledge, threatened against SM Energy.
- 3.5 Foreign Person. SM Energy is not a "disregarded entity" or a "foreign person" within the meaning of Section 1445 of the Code and its implementing Treasury Regulations.

7

3.6 Litigation. Except as set forth in Schedule 3.6, as of the Execution Date, there are no investigations, suits, actions or litigation by any Person by or before any Governmental Authority, and no legal, administrative or arbitration proceedings, pending or, to SM Energy's Knowledge, threatened in writing against SM Energy with respect to its ownership or operation of the Assets.

3.7 Material Contracts.

- (a) Schedule 3.7(a) sets forth all Applicable Contracts of the type described below (collectively, the 'Material Contracts'):
- (i) any Applicable Contract that can reasonably be expected to result in aggregate payments by SM Energy of more than \$50,000 during the current or any subsequent fiscal year or \$100,000 in the aggregate over the term of such Applicable Contract (in each case, based solely on the terms thereof and without regard to any expected increase in volumes or revenues) that cannot be terminated by SM Energy on not greater than 90 days' notice;
- (ii) any Applicable Contract that can reasonably be expected to result in aggregate revenues to SM Energy of more than \$50,000 during the current or any subsequent fiscal year or \$100,000 in the aggregate over the term of such Applicable Contract (in each case, based solely on the terms thereof and without regard to any expected increase in volumes or revenues);
- (iii) any Applicable Contract that (A) is a Hydrocarbon purchase and sale, gathering, transportation, processing, compression or similar Contract pursuant to which SM Energy received annual revenues or makes annual payments in excess of \$100,000 and (B) is not terminable by SM Energy or its assignee without penalty on 60 days' or less notice;
- (iv) any Applicable Contract that is an indenture, mortgage, loan, credit or sale-leaseback, guaranty of any obligation, bonds, letters of credit or similar financial Contract, except any such Applicable Contract with an aggregate outstanding principal amount not exceeding \$100,000;
- (v) any Applicable Contract that constitutes a lease under which SM Energy is the lessor or the lessee of real, immovable, personal or movable property which lease (A) cannot be terminated by SM Energy without penalty upon 60 days or less notice and (B) involves an annual base rental of more than \$50,000; and
- (vi) Applicable Contracts with any Affiliate of SM Energy that will be binding on Buyer after the Closing Date and will not be terminable by Buyer within 30 days' or less notice other than joint operating agreements.
- (b) There exist no material defaults under the Material Contracts by SM Energy or, to SM Energy's Knowledge, by any other Person that is a party to such Material Contracts.

- 3.8 No Violation of Laws. Except as set forth on Schedule 3.8, as of the Execution Date, to SM Energy's Knowledge, SM Energy and its Affiliates are not in material violation of any applicable Laws with respect to the ownership and operation (where applicable) by SM Energy of the Assets. This Section 3.8 does not include any matters with respect to Environmental Laws, such matters being addressed exclusively in Section 3.12.
- 3.9 Royalties, Etc. Except (A) for the Burdens under dispute as set forth on Schedule 3.9, (B) as would not have a Material Adverse Effect or (C) for such items that are being held in suspense as permitted pursuant to applicable Law, SM Energy has paid, and SM Energy has not received written notice from any third-party operator that such operator has not paid, all Burdens due by SM Energy with respect the Assets or, if SM Energy is responsible for paying such Burdens but has not paid any such Burdens, SM Energy is contesting such unpaid Burdens in good faith.
 - 3.10 Imbalances. Schedule 3.10 sets forth all Imbalances associated with the Assets as of the dates indicated on such Schedule.
- 3.11 Current Commitments. Schedule 3.11 sets forth, as of the Execution Date, all authorities for expenditures relating to the Assets to drill or rework Wells ("AFEs") for which all of the activities anticipated in such AFEs have not been completed by the Execution Date, and which require the owner of the Assets to expend in excess of \$50,000 after the Effective Time.

3.12 Environmental.

- (a) With respect to the Assets, SM Energy has not entered into, and is not subject to, any agreement, consent, order, decree, judgment, license or permit condition or other directive of any Governmental Authority that (i) are in existence as of the Execution Date, (ii) are based on any Environmental Laws that relate to the future use of any of the Assets and (iii) require any change in the present conditions of any of the Assets.
- (b) Except as set forth in *Schedule 3.12*, as of the Execution Date, SM Energy has not received written notice from any Person of any release, disposal, event, condition, circumstance, activity, practice or incident concerning any land, facility, asset or property included in the Assets that: (i) is in material violation of, interferes with or prevents material compliance by SM Energy with any Environmental Law or the terms of any license or permit issued pursuant thereto or (ii) gives rise to or results in any common Law or other liability of SM Energy to any Person.
- 3.13 Asset Taxes. Except as disclosed in Schedule 3.13, all Asset Taxes that have become due and payable by SM Energy prior to the Effective Time have been properly paid other than any Asset Taxes that are being contested in good faith.
- 3.14 Brokers' Fees. SM Energy has incurred no liability, contingent or otherwise, for brokers' or finders' fees relating to the transactions contemplated by this Agreement for which Buyer or any Affiliate of Buyer shall have any responsibility.
- 3.15 Advance Payments. SM Energy is not obligated by virtue of any take-or-pay payment, advance payment or other similar payment (other than royalties, overriding royalties

9

and similar arrangements reflected with respect to the Net Revenue Interests for the Wells set forth in **Exhibit B** and gas balancing arrangements reflected on *Schedule 3.10*), to deliver Hydrocarbons attributable to the Assets, or proceeds from the sale thereof, at some future time without receiving payment therefore at or after the time of delivery.

- 3.16 Partnerships. Schedule 3.16 sets forth all of the Assets that are deemed by agreement or applicable Law to be held by a partnership for federal Tax purposes and, to the extent any of the Assets are deemed by agreement or applicable Law to be held by a partnership for federal Tax purposes.
- 3.17 Payout Balances. Schedule 3.17 identifies the Wells and Units subject to a reversion or other adjustment at some level of cost recovery or payout (or passage of time or other event other than a termination of a Lease by its terms), and, to SM Energy's Knowledge, the list is a complete and accurate list of the status of any "payout" balance, as of the Effective Time.
- 3.18 Preferential Rights to Purchase and Consents. To SM Energy's Knowledge, Schedule 6.4 accurately lists all Assets subject to a preferential right to purchase or to a consent to assignment.
- 3.19 No Other Representations or Warranties; Disclosed Materials. Except for the representations and warranties contained in this Agreement (as qualified by the Schedules), neither SM Energy nor any other Person makes (and Buyer is not relying upon) any other express or implied representation or warranty with respect to SM Energy (including the value, condition or use of any of the Assets) or the transactions contemplated by this Agreement, and SM Energy disclaims any other representations or warranties not contained in this Agreement, whether made by SM Energy, any Affiliate of SM Energy, or any of their respective officers, directors, managers, employees or agents. Except for the representations and warranties contained in this Agreement (as qualified by the Schedules), SM Energy disclaims all liability and responsibility for any representation, warranty, projection, forecast, statement or information made, communicated or furnished (orally or in writing) to Buyer or any of its Affiliates or any of its officers, directors, managers, employees or agents (including any opinion, information, projection or advice that may have been or may be provided to Buyer by any director, officer, employee, agent, consultant or representative of SM Energy, or any of its Affiliates). The disclosure of any matter or item in the Schedules shall not be deemed to constitute an acknowledgement that any such matter is required to be disclosed or is material or that such matter would or would reasonably be expected to result in a Material Adverse Effect.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF BUYER

Each Buyer, severally and not jointly, represents and warrants to SM Energy, with respect to itself only, the following:

4.1 Organization; Existence. Buyer is a limited partnership duly formed and validly existing under the Laws of Delaware. Buyer has all requisite power and authority to own and operate its property (including, at Closing, the Assets) and to carry on its business as now

10

conducted. By Closing, Buyer will be duly licensed or qualified to do business as a foreign limited partnership and is in good standing in all jurisdictions in which such qualification is required by Law except where the failure to qualify or be in good standing would not have a material adverse effect upon the ability of Buyer to consummate the transactions contemplated by this Agreement.

4.2 Authorization. Buyer has full power and authority to enter into and perform this Agreement to which it is a party and the transactions contemplated herein. The execution, delivery and performance by Buyer of this Agreement has been, and the execution, delivery and performance by Buyer of all other documents delivered

pursuant to this Agreement will be when delivered, duly and validly authorized and approved by all necessary limited partnership action on the part of Buyer. Assuming the due authorization, execution and delivery by the other parties to such documents, this Agreement constitutes, and the other documents delivered pursuant to this Agreement to which Buyer is a party will constitute, Buyer's legal, valid and binding obligations, enforceable against Buyer in accordance with their respective terms, subject however to the effects of bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium and similar Laws relating to or affecting creditors' rights, as well as to principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at Law).

- 4.3 No Conflicts. The execution, delivery and performance by Buyer of this Agreement and the consummation of the transactions contemplated herein will not (a) conflict with or result in a breach of any provisions of the organizational or other governing documents of Buyer, (b) result in a default or the creation of any Encumbrance or give rise to any right of termination, cancellation or acceleration under any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license or other agreement to which Buyer is a party or by which Buyer or any of its property may be bound or (c) violate any Law applicable to Buyer or any of its property, except in the case of clauses (b) and (c) where such default, Encumbrance, termination, cancellation, acceleration or violation would not have a material adverse effect upon the ability of Buyer to consummate the transactions contemplated by this Agreement.
- 4.4 Consents. There are no consents or other restrictions on assignment, including requirements for consents from third parties to any assignment, in each case, that would be applicable in connection with the consummation by Buyer of the transactions contemplated by this Agreement.
- 4.5 Bankruptcy. There are no bankruptcy, reorganization or receivership proceedings pending, being contemplated by or, to Buyer's knowledge, threatened against Buyer.
- 4.6 Litigation. There is no investigation, suit, action or litigation by any Person or by or before any Governmental Authority, and no legal, administrative or arbitration proceedings pending, or to Buyer's knowledge, threatened in writing against Buyer, or to which Buyer is a party, that would affect the ability of Buyer to consummate the transactions contemplated by this Agreement.
- 4.7 Financing. Buyer has, and shall have as of the Closing Date, sufficient funds with which to pay the Closing Amount and consummate the transactions contemplated by this

11

Agreement and, following Closing, Buyer will have sufficient funds to pay any adjustments to the Purchase Price and meet its other payment obligations under this Agreement.

- 4.8 Independent Evaluation. Buyer is sophisticated in the evaluation, purchase, ownership and operation of oil and gas properties and related facilities. In making its decision to enter into this Agreement and to consummate the transaction contemplated herein, Buyer (a) has relied or shall rely solely on its own independent investigation and evaluation of the Assets and the advice of its own legal, tax, economic, environmental, engineering, geological and geophysical advisors and the express provisions of this Agreement and not on any comments, statements, projections or other materials made or given by any representatives or consultants or advisors engaged by SM Energy, and (b) has satisfied or shall satisfy itself through its own due diligence as to the environmental and physical condition of and contractual arrangements and other matters affecting the Assets.
- 4.9 Brokers' Fees. Buyer has incurred no liability, contingent or otherwise, for brokers' or finders' fees relating to the transactions contemplated by this Agreement for which SM Energy or any Affiliate of SM Energy shall have any responsibility.
- 4.10 Accredited Investor. Buyer is an "accredited investor," as such term is defined in Regulation D of the Securities Act of 1933, as amended, and will acquire the Assets for its own account and not with a view to a sale or distribution thereof in violation of such Law and the rules and regulations thereunder, any applicable state blue sky Laws or any other applicable securities Laws.

ARTICLE V ACCESS / DISCLAIMERS

5.1 Access.

From and after the Execution Date and up to and including the Closing Date (or earlier termination of this Agreement), but subject to the other provisions of this Agreement (including this Section 5.1) and obtaining any required consents of third parties (with respect to which consents SM Energy shall use its commercially reasonable efforts to obtain), SM Energy shall afford, or use commercially reasonable efforts to cause operators of the Assets to permit, to Buyer, its Affiliates and its and their officers, employees, agents, accountants, attorneys, investment bankers, consultants and other authorized representatives (collectively, "Buyer's Representatives") reasonable access, during normal business hours, to the Assets and all Records in SM Energy's or any of its Affiliates' possession. SM Energy shall also make available to Buyer and Buyer's Representatives, upon reasonable notice during normal business hours, SM Energy's personnel knowledgeable with respect to the Assets in order that Buyer may make such diligence investigation as Buyer considers necessary or appropriate. All investigations and due diligence conducted by Buyer or any Buyer's Representative shall be conducted at Buyer's sole cost, risk and expense; and any conclusions made from any examination done by Buyer or any Buyer's Representative shall result from Buyer's own independent review and judgment. Buyer shall coordinate its access rights and physical inspections of the Assets with SM Energy and any third party that serves as an Operator (each a "Third Party Operator") to reasonably minimize any inconvenience to or interruption of the conduct of business by SM

12

Energy or any Third Party Operator. Buyer shall give SM Energy reasonable prior written notice before entering onto any of the Assets so that SM Energy can arrange any on-site visit of the Assets with its employees and the third-person operator, if applicable. SM Energy shall have the right to have its representatives present at any time any Buyer's Representative is present on the Assets. Buyer shall, and shall cause all of the Buyer's Representatives to, abide by SM Energy's and any Third Party Operator's safety rules, regulations and operating policies while conducting its due diligence evaluation of the Assets including any environmental or other inspection or assessment of the Assets. Buyer agrees to coordinate all access with, and deliver any notice required pursuant to this *Section 5.1(a)* to:

Hans Swolfs Landman SM Energy Company 6120 S. Yale Ave. Suite 1300 Tulsa, OK 74136 Tel.: 918.877.4525

Email: hswolfs@sm-energy.com

(b) Buyer shall not conduct any sampling, boring, drilling or other invasive investigation activities ('Invasive Activities'') on or with respect to any of

the Assets without (i) SM Energy's prior written consent, which consent may not be unreasonably withheld and (ii) any required prior written consent from any third party, including any third-person operator. SM Energy shall, with the reasonable cooperation of Buyer and in reasonable consultation with Buyer, use commercially reasonable efforts to obtain any such required third party consent.

- Buyer agrees to defend, indemnify and hold harmless each of the Operators of the Assets and the SM Indemnified Parties from and against any and all Liabilities arising out of, resulting from or relating to any field visit, environmental property assessment or other due diligence activity conducted by Buyer or any Buyer's Representative (including an Invasive Activity, if any) with respect to the Assets, EVEN IF SUCH LIABILITIES ARISE OUT OF OR RESULT FROM, SOLELY OR IN PART, THE SOLE, ACTIVE, PASSIVE, CONCURRENT OR COMPARATIVE NEGLIGENCE, STRICT LIABILITY OR OTHER FAULT OR VIOLATION OF LAW OF OR BY A MEMBER OF THE SM INDEMNIFIED PARTIES, EXCEPTING ONLY LIABILITIES ACTUALLY RESULTING ON THE ACCOUNT OF THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF A MEMBER OF THE SM INDEMNIFIED PARTIES.
- (d) Buyer agrees to promptly provide SM Energy with copies, but in any event within 48 hours after its receipt or creation thereof, of all final reports and test results prepared by Buyer, any of Buyer's Representatives or any third party consultants and which contain data collected or generated from Buyer's due diligence with respect to the Assets (including an Invasive Activity, if any). SM Energy shall not be deemed by its receipt of said documents or otherwise to have made any representation or warranty, expressed, implied or statutory as to the condition to the Assets or to the accuracy of said documents or the information contained therein.

13

- (e) Upon completion of Buyer's due diligence, Buyer shall, at its sole cost and expense and without any cost or expense to SM Energy or its Affiliates (i) repair all damage done to the Assets in connection with Buyer's due diligence, (ii) restore the Assets to the approximate same or better condition in existence prior to commencement of Buyer's due diligence and (iii) remove all equipment, tools or other property brought onto the Assets in connection with Buyer's due diligence. Any disturbance to the Assets (including, without limitation, the real property associated with such Assets) resulting from Buyer's due diligence will be promptly corrected by Buyer.
- (f) During all periods that Buyer and/or any of Buyer's Representatives are present on the Assets, Buyer shall maintain, at its sole expense and with insurers reasonably satisfactory to SM Energy, Buyer's policies of insurance that are in place as of the Execution Date. Each such insurance policy required to be carried by Buyer hereunder will (i) be primary insurance, (ii) list the SM Indemnified Parties as additional insureds, (iii) waive subrogation against the SM Indemnified Parties and (iv) provide for five days prior notice to SM Energy in the event of cancellation or modification of such policy or reduction in the coverage of such policy. Buyer shall provide evidence of such insurance to SM Energy prior to entering the Assets
- 5.2 Confidentiality. Buyer acknowledges that, pursuant to its right of access to the Records and the Assets, Buyer will become privy to confidential and other information of SM Energy and that such confidential information shall be held confidential by Buyer and Buyer's Representatives in accordance with the terms of the Confidentiality Agreement. If Closing should occur, the foregoing confidentiality restriction on Buyer, including the Confidentiality Agreement, shall terminate (except as to (a) any Assets that are excluded from the transactions contemplated hereby pursuant to the provisions of this Agreement, (b) the Excluded Assets, and (c) information related to assets other than the Assets).

5.3 Disclaimers.

- (a) EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN *ARTICLE III*, (I) SM ENERGY MAKES NO REPRESENTATIONS OR WARRANTIES, EXPRESS, STATUTORY OR IMPLIED, AND (II) SM ENERGY EXPRESSLY DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, STATEMENT OR INFORMATION MADE OR COMMUNICATED (ORALLY OR IN WRITING) TO BUYER OR ANY BUYER'S REPRESENTATIVE (INCLUDING, WITHOUT LIMITATION, ANY OPINION, INFORMATION, PROJECTION OR ADVICE THAT MAY HAVE BEEN PROVIDED TO BUYER BY ANY OFFICER, DIRECTOR, EMPLOYEE, AGENT, CONSULTANT, REPRESENTATIVE OR ADVISOR OF SM ENERGY OR ANY OF ITS AFFILIATES).
- (b) EXCEPT AS EXPRESSLY SET FORTH IN *ARTICLE III*, AND WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, SM ENERGY EXPRESSLY DISCLAIMS ANY REPRESENTATION OR WARRANTY, EXPRESS, STATUTORY OR IMPLIED, AS TO (I) TITLE TO ANY OF THE ASSETS, (II) THE CONTENTS, CHARACTER OR NATURE OF ANY REPORT OF ANY PETROLEUM ENGINEERING CONSULTANT, OR ANY ENGINEERING, GEOLOGICAL OR SEISMIC

14

DATA OR INTERPRETATION, RELATING TO THE ASSETS, (III) THE QUANTITY, QUALITY OR RECOVERABILITY OF HYDROCARBONS IN OR FROM THE ASSETS, (IV) ANY ESTIMATES OF THE VALUE OF THE ASSETS OR FUTURE REVENUES GENERATED BY THE ASSETS, (V) THE ability to produce HYDROCARBONS FROM THE ASSETS, (VI) THE MAINTENANCE, REPAIR, CONDITION, QUALITY, SUITABILITY, DESIGN OR MARKETABILITY OF THE ASSETS, (VII) THE CONTENT, CHARACTER OR NATURE OF ANY INFORMATION MEMORANDUM, REPORTS, BROCHURES, CHARTS OR STATEMENTS PREPARED BY SM ENERGY OR THIRD PARTIES WITH RESPECT TO THE ASSETS, (VII) ANY OTHER MATERIALS OR INFORMATION THAT MAY HAVE BEEN MADE AVAILABLE TO BUYER OR ANY BUYER'S REPRESENTATIVE IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY DISCUSSION OR PRESENTATION RELATING THERETO AND (IX) ANY IMPLIED OR EXPRESS WARRANTY OF FREEDOM FROM PATENT OR TRADEMARK INFRINGEMENT. EXCEPT AS EXPRESSLY REPRESENTED OTHERWISE IN ARTICLE III, SM ENERGY FURTHER DISCLAIMS ANY REPRESENTATION OR WARRANTY, EXPESS, STATUTORY OR IMPLIED, OF MERCHANTABILITY, FREEDOM FROM LATENT VICES OR DEFECTS, FITNESS FOR A PARTICULAR PURPOSE OR CONFORMITY TO MODELS OR SAMPLES OF MATERIALS OF ANY ASSETS, RIGHTS OF A PURCHASER UNDER APPROPRIATE STATUTES TO CLAIM DIMINUTION OF CONSIDERATION OR RETURN OF CONSIDERATION, IT BEING EXPRESSLY UNDERSTOOD AND AGREED BY THE PARTIES THAT BUYER SHALL BE DEEMED TO BE OBTAINING THE ASSETS IN THEIR PRESENT STATUS, CONDITION AND STATE OF REPAIR, "AS IS" AND "WHERE IS" WITH ALL FAULTS OR DEFECTS (KNOWN OR UNKNOWN, LATENT, DISCOVERABLE OR UNDISCOVERABLE), AND THAT BUYER HAS MADE OR CAUSED TO BE MADE SUCH INSPECTIONS AS BUYER DEEMS APPROPRIATE.

- (c) OTHER THAN THOSE REPRESENTATIONS SET FORTH IN SECTION 3.12, SM ENERGY HAS NOT AND WILL NOT MAKE ANY REPRESENTATION OR WARRANTY REGARDING ANY MATTER OR CIRCUMSTANCE RELATING TO ENVIRONMENTAL LAWS, THE RELEASE OF MATERIALS INTO THE ENVIRONMENT OR THE PROTECTION OF HUMAN HEALTH, SAFETY, NATURAL RESOURCES OR THE ENVIRONMENT, OR ANY OTHER ENVIRONMENTAL CONDITION OF THE ASSETS, AND NOTHING IN THIS AGREEMENT OR OTHERWISE SHALL BE CONSTRUED AS SUCH A REPRESENTATION OR WARRANTY, AND SUBJECT TO BUYER'S limited RIGHTS UNDER SECTION 7.1, BUYER SHALL BE DEEMED TO BE TAKING THE ASSETS "AS IS" AND "WHERE IS" WITH ALL FAULTS FOR PURPOSES OF THEIR ENVIRONMENTAL CONDITION AND THAT BUYER HAS MADE OR CAUSED TO BE MADE SUCH ENVIRONMENTAL INSPECTIONS AS BUYER DEEMS APPROPRIATE.
- (d) SM ENERGY and Buyer agree that, to the extent required by applicable law to be effective, the disclaimers of certain representations and warranties contained in this Section 5.3 are "conspicuous" disclaimers for the purpose of any applicable Law.

ARTICLE VI TITLE MATTERS: CASUALTIES

6.1 SM Energy's Title. Without limiting Buyer's remedies for Title Defects set forth in this Article VI, SM Energy makes no warranty or representation, express, implied, statutory or otherwise with respect to its title to any of the Assets and Buyer hereby acknowledges and agrees that, except for the special warranty of title contained in the Assignment, Buyer's sole remedy for any defect of title, including any Title Defect, with respect to any of the Assets shall be as set forth in Section 6.2.

6.2 Notice of Title Defects; Defect Adjustments.

- (a) <u>Title Defect Notices.</u> On or before 4:00 p.m. (Mountain Time) on December 23, 2013 (the "Defect Claim Date"), Buyer shall have the right, but not the obligation, to deliver notices to SM Energy meeting the requirements of this Section 6.2(a) (each, a "Title Defect Notice") setting forth any matters which, in Buyer's reasonable opinion, constitute Title Defects and which Buyer asserts as a Title Defect pursuant to this Section 6.2. For all purposes of this Agreement and notwithstanding anything herein to the contrary, Buyer shall be deemed to have waived, and SM Energy shall have no liability for, any Title Defect that Buyer fails to assert as a Title Defect pursuant to a Title Defect Notice delivered in compliance with this Section 6.2(a) and received by SM Energy on or before the Defect Claim Date. To be effective, each Title Defect Notice shall be in writing and shall include (i) a description of the alleged Title Defect and the Assets affected by such Title Defect (each a "Title Defect Property"), (ii) the Allocated Value of each Title Defect Property is reduced by the existence of the alleged Title Defect(s), (iv) the amount by which Buyer reasonably believes the Allocated Value of each Title Defect Property is reduced by the alleged Title Defect(s), and (v) the computations upon which Buyer's belief is based. To give SM Energy an opportunity to commence reviewing and curing Title Defects, Buyer agrees to use reasonable efforts to give SM Energy, on or before the end of each calendar week prior to the Defect Claim Date, written notice of all alleged Title Defect Claim Date.
- (b) <u>Title Benefit Notices.</u> SM Energy shall have the right, but not the obligation, to deliver to Buyer on or before the Defect Claim Date, a notice setting forth any matters that, in SM Energy's reasonable opinion, constitute Title Benefits and that SM Energy asserts as a Title Benefit pursuant to this *Section 6.2* (each, a "*Title Benefit Notice*") including (i) a description of the Title Benefit and the Assets affected by the Title Benefit (the '*Title Benefit Property*"), (ii) the Allocated Value of the Title Benefit Property, (iii) supporting documents reasonably for Buyer to verify the existence of the alleged Title Benefit, (iv) the amount by which SM Energy reasonably believes the Allocated Value of the Title Benefit Property is increased by the Title Benefit and (v) the computations upon which SM Energy's belief is based. SM Energy shall be deemed to have waived all Title Benefits of which it has not given notice on or before the Defect Claim Date.
- (c) <u>Remedies for Title Defects.</u> Subject to SM Energy's continuing right to dispute the existence of a Title Defect or the Title Defect Amount asserted with respect thereto

16

and subject to the Individual Title Defect Threshold and the Aggregate Deductible, in the event that any Title Defect properly asserted by Buyer in accordance with Section 6.2(a) is not waived in writing by Buyer or cured on or before Closing, SM Energy shall, at its sole option, elect to:

- (i) transfer, convey and assign the entirety of the Title Defect Property that is subject to such Title Defect, together with all associated Assets, to Buyer at Closing, and reduce the Purchase Price by the Title Defect Amount; or
- (ii) cure such Title Defect Property that is subject to such Title Defect, in which event SM Energy shall have the right, for a period of 110 days following the Closing Date (such period, the "Cure Period"), to cure any Title Defect relating to such Title Defect Property.

If SM Energy elects to cure any Title Defect pursuant to clause (ii) of this Section 6.2, Buyer shall withhold from the Purchase Price the Allocated Value of such Cure Assets. At Closing, (i) SM Energy shall remove from the Assignment any Asset or portions thereof subject to such Title Defect, as applicable, which SM Energy has elected to cure (the "Cure Assets") and with respect to which Buyer has retained the applicable Allocated Value and (ii) if the applicable Title Defect is cured as provided herein or determined to have been cured as provided herein within the Cure Period, within five (5) Business Days after the expiration of the Cure Period, SM Energy shall convey the applicable Cure Assets to Buyer and Buyer shall pay to SM Energy the applicable Allocated Value less any Title Defect Amount related to any remaining Title Defect(s) applicable to such Cure Assets, on the same terms as the Assets conveyed at Closing. During the Cure Period, and regardless of whether SM Energy cures a Title Defect affecting Cure Asset(s) retained pursuant to this Section 6.2(c), with respect to any Cure Asset, SM Energy shall have the option to elect the remedy for such Title Defect set forth in Section 6.2(c)(i), in which event SM Energy shall transfer, convey and assign such Cure Asset, together with all associated Assets, to Buyer, and Buyer shall pay to SM Energy the Allocated Value less the Title Defect Amount related to the uncured Title Defect(s) applicable to such Cure Assets, on the same terms as the Assets conveyed at Closing.

- (d) <u>Remedies for Title Benefits.</u> With respect to each Well or Well Location affected by Title Benefits reported under this *Section 6.2*, the aggregate Title Defect Amounts shall be decreased by an amount equal to the increase in the Allocated Value for such Asset caused by such Title Benefits, as determined pursuant to *Section 6.2(g)* (the "*Title Benefit Amount*").
- (e) Exclusive Remedy. Except for the special warranty of title contained in the Assignment, the provisions set forth in Section 6.2(c) and Section 6.2(i) shall be the exclusive right and remedy of Buyer with respect to SM Energy's failure to have Defensible Title with respect to any Asset. Buyer agrees that after Closing, it will not assert a special warranty of title claim for any Title Defect it discovered prior to the Defect Claim Date.
- (f) <u>Title Defect Amount</u>. The amount by which the Allocated Value of the affected Title Defect Property is reduced as a result of the existence of such Title Defect shall be

17

the "Title Defect Amount" and shall be determined in accordance with the following terms and conditions:

- (i) if Buyer and SM Energy agree on the Title Defect Amount, then that amount shall be the Title Defect Amount;
- (ii) if the Title Defect is an Encumbrance that is liquidated in amount, then the Title Defect Amount shall be the amount necessary to be paid to remove the Title Defect from the Title Defect Property;
- (iii) if the Title Defect represents a discrepancy between (A) the Net Revenue Interest for any Title Defect Property and (B) the Net Revenue Interest for such Title Defect Property stated in *Schedule 2.5*, and the Working Interest attributable to such Title Defect Property has been reduced proportionately, then the Title Defect Amount shall be the product of the Allocated Value of such Title Defect Property multiplied by a fraction, the numerator of which is the positive difference between such Net Revenue Interest values, and the denominator of which is the Net Revenue Interest for such Title Defect Property

stated in Schedule 2.5;

- (iv) if the Title Defect represents an obligation or Encumbrance upon or other defect in title to the Title Defect Property of a type not described above, including a Title Defect that represents a discrepancy between (A) the Net Revenue Interest for any Title Defect Property and (B) the Net Revenue Interest for such Title Defect Property stated in Schedule 2.5, and the Working Interest attributable to such Title Defect Property has not been reduced proportionately, the applicable Title Defect Amount shall be determined by taking into account the Allocated Value of the Title Defect Property, the portion of the Title Defect Property affected by the Title Defect, the legal effect of the Title Defect, the potential economic effect of the Title Defect over the life of the Title Defect Property, the values placed upon the Title Defect by Buyer and SM Energy and such other reasonable factors as are necessary to make a proper evaluation; provided, however, that if such Title Defect is reasonably capable of being cured, the Title Defect Amount shall not be greater than the lesser of (X) the reasonable cost and expense of curing such Title Defect and (Y) the Allocated Value of such Title Defect Property;
- (v) the Title Defect Amount with respect to a Title Defect Property shall be determined without duplication of any costs or losses included in any other Title Defect Amount hereunder; and
- (vi) notwithstanding anything to the contrary in this *Article VI*, the aggregate Title Defect Amounts attributable to the effects of all Title Defects upon any single Title Defect Property shall not exceed the Allocated Value of such Title Defect Property.
- (g) <u>Title Benefit Amount</u>. The Title Benefit Amount resulting from a Title Benefit shall be determined in accordance with the following methodology, terms and conditions:

18

- (i) if Buyer and SM Energy agree on the Title Benefit Amount, then that amount shall be the Title Benefit Amount;
- (ii) if the Title Benefit represents an increase between (A) the Net Revenue Interest for any Title Benefit Property and (B) the Net Revenue Interest for such Title Benefit Property stated in *Schedule 2.5*, and the Working Interest has increased proportionately, then the Title Benefit Amount shall be the product of the Allocated Value of such Title Benefit Property multiplied by a fraction, the numerator of which is the positive difference between such Net Revenue Interest values, and the denominator of which is the Net Revenue Interest for such Title Benefit Property stated in *Schedule 2.5*, provided that if the Net Revenue increase does not affect the Title Benefit Property throughout its entire life, the Title Benefit Amount determined under this *Section 6.2(g)* shall be reduced to take into account the applicable time period only;
- (iii) if the Title Benefit is of a type not described above, then the applicable Title Benefit Amount shall be determined by taking into account the Allocated Value of the Title Benefit Property, the portion of the Title Benefit Property affected by such Title Benefit, the legal effect of the Title Benefit, the potential economic effect of the Title Benefit over the life of such Title Benefit Property, the values placed upon the Title Benefit by Buyer and SM Energy and such other reasonable factors as are necessary to make a proper evaluation; and
- (iv) the Title Benefit Amount with respect to a Title Benefit Property shall be determined without duplication of any items or amounts included in any other Title Benefit Amount hereunder.
- (h) Title Threshold and Deductibles. Notwithstanding anything to the contrary, in no event shall there be any adjustments to the Purchase Price or other remedies provided by SM Energy hereunder: (i) for any individual Title Defect for which the Title Defect Amount does not exceed the Individual Title Defect Threshold and (ii) for any Title Defect for which the Title Defect Amount exceeds the Individual Title Defect Threshold unless the sum of (A) the aggregate Title Defect Amounts of all such Title Defects that exceed the Individual Title Defect Amounts attributable to Title Defects cured by SM Energy), plus (B) the aggregate Remediation Amounts of all Environmental Defects that exceed the Individual Environmental Threshold (excluding any Remediation Amounts attributable to (1) Environmental Defects cured by SM Energy or (2) Environmental Defect Properties retained by SM Energy pursuant to Section 7.1(b)(ii)), exceeds [***] of the unadjusted Purchase Price (the "Aggregate Deductible"), after which point Buyer shall be entitled to adjustments to the Purchase Price only with respect to such Title Defects in excess of the Aggregate Deductible. For the avoidance of doubt, if SM Energy elects to cure any Title Defect pursuant to Section 6.2(c)(ii), the Title Defect Amount relating to such Title Defect Property will not be counted toward the Aggregate Deductible, unless SM Energy is unable to cure such Title Defect during the Cure Period
- (i) <u>Title Dispute Resolution.</u> The Parties agree to resolve disputes concerning the following matters pursuant to this *Section 6.2(i)*: (1) the existence and scope of a Title Defect or Title Defect Amount, (2) the adequacy of SM Energy's Title Defect curative materials and

19

Buyer's reasonable satisfaction thereof, and (3) the existence and scope of a Title Benefit or Title Benefit Amount (the "Title Disputed Matters"). The Parties agree to attempt to initially resolve all disputes through good faith negotiations. If the Parties cannot resolve disputes regarding items (1) and (2) on or before Closing, the Closing shall be delayed as to only the Assets subject to the Title Disputed Matters until the Parties finally resolve the dispute pursuant to this Section 6.2(i); provided, however, if either Party asserts that the condition in Section 9.1(d) or Section 9.2(d) has not been satisfied due, in whole or in part, to Title Defects, then the Parties will resolve all Title Disputed Matters pursuant to this Section 6.2(i) prior to Closing. In the event that neither Party asserts that the condition in Section 9.1(d) or Section 9.2(d) has not been satisfied, it is understood and agreed that the Parties shall proceed to Closing as contemplated herein as to all Assets not covered by a Title Disputed Matter. The Title Disputed Matters will be finally determined pursuant to this Section 6.2(i). There shall be a single arbitrator, who shall be an attorney with at least 10 years' experience in oil and gas title and transaction matters, as selected by mutual agreement of Buyer and SM Energy within 15 days after any Party invokes the provisions of this Section 6.2(i) to resolve such Dispute, and absent such agreement, by the Houston office of the AAA (the "Title Arbitrator"). The arbitration proceeding shall be held in Houston, Texas and shall be conducted in accordance with the AAA Rules to the extent such rules do not conflict with the terms of this Section 6.2(i). The Title Arbitrator's determination shall be made within 20 days after submission by the Parties of the matters in Dispute and shall be final and binding upon both Parties, without right of appeal. In making his determination, the Title Arbitrator shall be bound by the rules set forth in Section 6.2(f) and Section 6.2(g) and, subject to the foregoing, may consider such other matters as in the opinion of the Title Arbitrator are necessary to make a proper determination. The Title Arbitrator, however, may not award (a) Buyer a greater Title Defect Amount than the Title Defect Amount claimed by Buyer in the applicable Title Defect Notice (which such Title Defect Amount shall not exceed the Allocated Value of the applicable Title Defect Property) or (b) SM Energy a greater Title Benefit Amount than the Title Benefit Amount claimed by SM Energy in the applicable Title Benefit Notice. The Title Arbitrator shall act as an expert for the limited purpose of determining the specific disputed Title Defect, Title Benefit, Title Defect Amount or Title Benefit Amount submitted by either Party and may not award damages, interest or penalties to either Party with respect to any Dispute. SM Energy and Buyer shall each bear its own legal fees and other costs of presenting its case to the Title Arbitrator. Each of SM Energy and Buyer shall bear one-half of the costs and expenses of the Title Arbitrator. To the extent that the award of the Title Arbitrator with respect to any Title Defect Amount was not taken into account as an adjustment to the Purchase Price or the aggregate Title Defect Amounts, as applicable at Closing pursuant to Section 2.4 and an adjustment would otherwise be required under the provisions of Section 6.2(c) or Section 6.2(d), as applicable, then, within 10 days after the Title Arbitrator delivers written notice to Buyer and SM Energy of its award with respect to such Title Defect Amount or a Title Benefit Amount and subject to Section 6.2(h), the Purchase Price will be adjusted pursuant to Section 2.4 by the amount so awarded by the Title Arbitrator.

(a) Notwithstanding anything herein to the contrary, from and after the Effective Time, if Closing occurs, Buyer shall assume all risk of loss with respect to (i) production of Hydrocarbons from the Assets through normal depletion (including watering out of any well, collapsed casing or sand infiltration of any well) and (ii) the depreciation of personal

20

property due to ordinary wear and tear and, in each case, Buyer shall not assert such matters as Casualty Losses or Title Defects hereunder.

(b) If, from and after the Effective Time but prior to the Closing Date, any portion of the Assets is damaged or destroyed by fire or other casualty or is taken in condemnation or under right of eminent domain (a "Casualty Loss"), and the resulting loss from such Casualty Loss exceeds \$50,000 based on the Allocated Value of the affected Assets, then (i) Buyer shall nevertheless be required to close the transactions contemplated by this Agreement and (ii) SM Energy shall elect by written notice to Buyer prior to Closing to either (A) cause, at SM Energy's sole cost and as promptly as reasonably practicable (which work may extend after the Closing Date), each Asset affected by such Casualty Loss to be repaired or restored to at least its condition prior to such casualty or taking, or (B) reduce the Purchase Price by the cost to repair or restore each Asset affected by such Casualty Loss to at least its condition prior to such casualty or taking. In each case, SM Energy shall retain all rights to insurance, condemnation awards and other claims against third parties with respect to the casualty or taking except to the extent the Parties otherwise agree in writing.

6.4 Preferential Rights and Consents to Assign.

- (a) All consents to assign relating to the Assets ("Consents") and preferential rights to purchase ("Preferential Rights") are listed on Schedule 6.4. Other than in relation to a breach of the representation and warranty in Section 3.18, the remedies set forth in this Section 6.4 are the exclusive remedies under this Agreement related to the Consents and Preferential Rights.
- (b) From and after the Execution Date up to Closing, SM Energy and Buyer shall use their commercially reasonable efforts to obtain all Consents and waivers of all Preferential Rights (excluding any Customary Post-Closing Consents); provided, however, that neither Party shall be required to incur any Liability or pay any money in order to obtain such Consents or waivers.
- (c) <u>Preferential Rights</u>. SM Energy shall, within 10 days after the Execution Date, send to each holder of a Preferential Right a notice requesting the election or waiver by each such holder of its applicable Preferential Right, in each case in compliance with the contractual provisions applicable to such Preferential Right, requesting a waiver of such right. Any Preferential Right must be exercised subject to all terms and conditions set forth in this Agreement, including the successful closing of this Agreement pursuant to *Article XI*. The consideration payable under this Agreement for any particular Asset for purposes of Preferential Right notices shall be the Allocated Value of such Asset.
 - (i) All Assets burdened by Preferential Rights for which (A) the applicable Preferential Right has been waived, or (B) the period to exercise such Preferential Right has expired prior to the Closing without the applicable holder of such Preferential Right electing to enforce its Preferential Right, shall, in each case, be assigned to Buyer at the Closing pursuant to the provisions of this Agreement.

21

- (ii) If, prior to the Closing (A) any holder of a Preferential Right notifies SM Energy that it intends to consummate the purchase of the portion of the Assets to which its Preferential Right applies or (B) the time for exercising a Preferential Right has not expired and the holder of such Preferential Right has not waived such Preferential Right, then, in each case, such portion of the Assets affected by such Preferential Right shall be excluded from the Assets to be conveyed to Buyer at Closing and the Purchase Price shall be reduced by the Allocated Value of such excluded portion of the Assets. SM Energy shall be entitled to all proceeds paid by a Person exercising a Preferential Right prior to the Closing. If, after Closing (1) such holder of such Preferential Right thereafter fails to consummate the purchase of the portion of the Assets covered by such Preferential Right or (2) the time for exercising such Preferential Right expires without exercise by the holder thereof, then SM Energy shall (x) so notify Buyer and (y) on or before 10 days following delivery of such notice, assign such portion of the Assets to Buyer pursuant to an assignment in substantially the form of the Assignment and the Purchase Price shall be increased by an amount equal to the Allocated Value of the such portion of the Assets.
- (d) Consents. SM Energy, within 10 days after the Execution Date, shall send to each holder of a Consent a notice seeking such holder's consent to the transactions contemplated hereby.
 - (i) If (A) SM Energy fails to obtain a Hard Consent prior to Closing or (B) a Consent requested by SM Energy is denied in writing, then, in each case, that portion of the Assets affected by such Consent shall be excluded from the Assets to be conveyed to Buyer at Closing and the Purchase Price shall be reduced by the Allocated Value of such portion of the Assets. In the event that a Consent that was not obtained prior to Closing is obtained following Closing or the requirement to obtain such Consent is waived by Buyer then, within 10 days after such Consent is obtained or the requirement to obtain such Consent is waived by Buyer, (x) SM Energy shall assign such excluded portion of the Assets to Buyer pursuant to an assignment in substantially the form of the Assignment (and if the requirement to obtain a Consent is waived by Buyer, Buyer shall have no claim against, and SM Energy shall have no Liability for, the failure to obtain such Consent), and (y) Buyer shall pay to SM Energy by wire transfer of immediately available funds an amount equal to the amount by the Allocated Value of such portion of the Assets so assigned.
 - (ii) If (A) SM Energy fails to obtain a Soft Consent prior to Closing and (B) such Consent requested by SM Energy is not denied in writing, then that portion of the Assets subject to such Consent shall be assigned by SM Energy to Buyer at Closing pursuant to the Assignment and Buyer shall have no claim against, and SM Energy shall have no Liability for, the failure to obtain such Consent.

22

ARTICLE VII ENVIRONMENTAL MATTERS

7.1 Environmental Defects.

Environmental Defects Notice. On or before the Defect Claim Date, Buyer shall have the right, but not the obligation, to deliver notices to SM Energy meeting the requirements of this Section 7.1(a) (each, an "Environmental Defect Notice") setting forth any matters that, in Buyer's reasonable opinion, constitute Environmental Defects and that Buyer asserts as Environmental Defects pursuant to this Section 7.1. For all purposes of this Agreement, but subject to Section 7.1(c) below, Buyer shall be deemed to have waived any Environmental Defect that Buyer fails to properly assert as an Environmental Defect pursuant to an Environmental Defect Notice delivered in accordance with this Section 7.1(a) and received by SM Energy on or before the Defect Claim Date. To be effective, each Environmental Defect Notice shall be in writing and shall include: (i) a description of the Environmental Condition constituting the alleged Environmental Defect, including the Environmental Law(s) violated by

such Environmental Defect, (ii) each Asset (or portion thereof) affected by the alleged Environmental Defect (the "Environmental Defect Property"), (iii) the Allocated Value of each Environmental Defect Property, (iv) supporting documents reasonably necessary for SM Energy to verify the existence of the alleged Environmental Defect, and (v) a calculation of the Remediation Amount (itemized in reasonable detail) that Buyer asserts is attributable to such alleged Environmental Defect. Buyer's calculation of the Remediation Amount included in the Environmental Defect Notice must describe in reasonable detail the Remediation proposed for the Environmental Condition that gives rise to the asserted Environmental Defect and identify all assumptions used by Buyer in calculating the Remediation Amount, including the standards that Buyer asserts must be met to comply with Environmental Laws. SM Energy shall have the right, but not the obligation, to cure any claimed Environmental Defect on or before Closing. To give SM Energy an opportunity to commence reviewing and curing Environmental Defects, Buyer agrees to use reasonable efforts to give SM Energy, on or before the end of each calendar week prior to the Defect Claim Date, written notice of all alleged Environmental Defects discovered by Buyer during the preceding calendar week, which notice may be preliminary in nature and supplemented prior to the expiration of the applicable Defect Claim Date.

- (b) <u>Remedies for Environmental Defects.</u> Subject to SM Energy's continuing right to dispute the existence of an Environmental Defect or the Remediation Amount asserted with respect thereto and subject to the Individual Environmental Threshold and the Aggregate Deductible, in the event that any Environmental Defect timely asserted by Buyer in accordance with *Section 7.1(a)* is not waived in writing by Buyer or cured on or before Closing, SM Energy shall, at its sole option, elect to:
 - (i) reduce the Purchase Price by the Remediation Amount for such Environmental Defect; or
 - (ii) (A) if the Remediation Amount for such Environmental Defect is less than the Allocated Value of such Environmental Defect Property and if agreed to by the Buyer, retain the entirety of the Environmental Defect Property subject to such Environmental Defect, together with all associated Assets, and reduce the

23

Purchase Price by an amount equal to the Allocated Value of the Environmental Defect Property and associated Assets; or (B) if the Remediation Amount for such Environmental Defect is greater than the Allocated Value of such Environmental Defect Property, retain the entirety of the Environmental Defect Property subject to such Environmental Defect, together with all associated Assets, and reduce the Purchase Price by an amount equal to the Allocated Value of the Environmental Defect Property and associated Assets.

If SM Energy elects the option set forth in clause (i) above, Buyer shall be deemed to have assumed responsibility for all costs and expenses attributable to the Remediation of the applicable Environmental Defect (net to the Assets) and all Liabilities (net to the Assets) with respect thereto, and Buyer's obligations with respect to the foregoing shall be deemed to constitute part of the Assumed Obligations.

- (c) <u>Exclusive Remedy</u>. Subject to Buyer's remedy for a breach of SM Energy's representation contained in *Section 3.12*, *Section 7.1(b)* shall be the exclusive right and remedy of Buyer with respect to any Environmental Defect.
- (d) Environmental Deductibles. Notwithstanding anything to the contrary, in no event shall there be any adjustments to the Purchase Price or other remedies provided by SM Energy for (i) any individual Environmental Defect for which the Remediation Amount does not exceed the Individual Environmental Threshold, or (ii) any Environmental Defect for which the Remediation Amount exceeds the Individual Environmental Threshold, unless the sum of (A) the aggregate Remediation Amounts of all such Environmental Defects that exceed the Individual Environmental Threshold (excluding any Remediation Amounts attributable to (1) Environmental Defects cured by SM Energy or (2) Environmental Defect Properties that SM Energy elects to retain pursuant to Section 7.1(b)(ii)), plus (B) the aggregate Title Defect Amounts of all Title Defects that exceed the Individual Title Defect Threshold (excluding any Title Defect Amounts attributable to Title Defects cured by SM Energy), exceeds the Aggregate Deductible, after which point Buyer shall only be entitled to adjustments to the Purchase Price with respect to such Environmental Defects in excess of the Aggregate Deductible. For the avoidance of doubt, if SM Energy elects to retain any Asset pursuant to Section 7.1(b)(ii), the Purchase Price shall be reduced by the Allocated Value of such retained Asset and the Remediation Amount for the Environmental Defect relating to such retained Asset will not be counted towards the Aggregate Deductible.
- (e) Environmental Dispute Resolution The Parties agree to resolve disputes concerning the existence and scope of an Environmental Defect or Remediation Amount pursuant to this Section 7.1(e) (the "Environmental Disputed Matters"). The Parties agree to attempt to initially resolve all disputes through good faith negotiations. If the Parties cannot resolve disputes regarding Environmental Disputed Matters on or before Closing, the Closing shall be delayed as to only the Assets subject to the Environmental Disputed Matters until the Parties finally resolve the dispute pursuant to this Section 7.1(e); provided, however, if either Party asserts that the condition in Section 9.2(d) has not been satisfied due, in whole or in part, to Environmental Defects, then the Parties will resolve all Environmental Disputed Matters pursuant to this Section 7.1(e) prior to Closing. In the event that neither Party asserts that the condition in Section 9.1(d) or Section 9.2(d) has not been satisfied, it is

24

understood and agreed that the Parties shall proceed to Closing as contemplated herein as to all Assets not covered by a Environmental Disputed Matter. The Environmental Disputed Matters will be finally determined pursuant to this Section 7.1(e). There will be a single arbitrator, who must be an environmental attorney with at least 10 years' experience in oil and gas environmental matters, as selected by mutual agreement of Buyer and SM Energy within 15 days after any Party invokes the provisions of this Section 7.1(e) to resolve such Dispute, and absent such agreement, by the Houston office of the AAA (the 'Environmental Arbitrator'). The arbitration proceeding will be held in Houston, Texas and conducted in accordance with the AAA Rules to the extent such rules do not conflict with the terms of this Section 7.1(e). The Environmental Arbitrator's determination must be made within 20 days after submission of the matters in Dispute and shall be final and binding upon both Parties, without right of appeal. In making its determination, the Environmental Arbitrator shall be bound by the rules set forth in this Section 7.1 and, subject to the foregoing, may consider such other matters as in the opinion of the Environmental Arbitrator are necessary or helpful to make a proper determination. The Environmental Arbitrator, however, may not award Buyer a greater Remediation Amount than the Remediation Amount claimed by Buyer in the applicable Environmental Defect Notice. The Environmental Arbitrator will act as an expert for the limited purpose of determining the specific disputed Environmental Defects and/or Remediation Amounts submitted by either Party and may not award damages, interest or penalties to either Party with respect to any matter. SM Energy and Buyer will each bear its own legal fees and other costs of presenting its case. Each Party will bear one-half of the costs and expenses of the Environmental Arbitrator. To the extent that the award of the Environmental Arbitrator with respect to any Remediation Amou

7.2 NORM, Wastes and Other Substances. Buyer acknowledges that the Assets have been used for exploration, development and production of oil and gas and that there may be petroleum, produced water, wastes or other substances or materials located in, on or under or associated with the Assets. Equipment and sites included in the Assets may contain asbestos, NORM or other Hazardous Substances. NORM may affix or attach itself to the inside of wells, materials and equipment as scale, or in other forms. The wells, materials and equipment located on the Assets or included in the Assets may contain NORM and other wastes or Hazardous Substances. NORM containing material or other wastes or Hazardous Substances may have come in contact with various environmental media, including without limitation, water, soils or sediment. Special procedures may be required for the assessment, remediation, removal, transportation or disposal of environmental media, wastes, asbestos, NORM and other Hazardous Substances from the Assets.

CERTAIN AGREEMENTS

8.1 Conduct of Business. Except (x) as set forth in Schedule 8.1, (y) as expressly contemplated by this Agreement or (z) as expressly consented to in writing by Buyer, SM Energy agrees that from and after the Execution Date up to Closing:

25

- (a) SM Energy will, and will cause its Affiliates to:
- (i) maintain, and if SM Energy is the Operator, operate, the Assets in the usual, regular and ordinary manner consistent with its past practice;

(ii) maintain the books of account and records relating to the Assets in the usual, regular and ordinary manner, in accordance with its usual accounting practices;

- (iii) give written notice to Buyer as soon as is practicable of any written notice received or given by SM Energy with respect to any alleged material breach by SM Energy of any Lease or Material Contract;
 - (iv) keep Buyer reasonably informed as to material events involving the Assets; and
- (v) consult with Buyer regarding operations involving the Assets and elections to be made under any Applicable Operating Agreement.
 - (b) SM Energy will not, and will cause its Affiliates not to:
- (i) except for (A) emergency operations, (B) operations required under presently existing AFEs described on *Schedule 3.11*, (C) operations undertaken to avoid any penalty provision of any Applicable Contract or Governmental Authority order, and (D) operations proposed by third parties relating to drilling, sidetracking, reworking or other similar operations with respect to the Assets operated by third parties, agree to, propose or commence any operations on the Assets anticipated to cost (net to the Assets) in excess of \$100,000 per operation; provided that with respect to emergency operations, SM Energy shall notify Buyer of such emergency as soon as reasonably practicable;
- (ii) enter into an Applicable Contract that, if entered into prior to the Execution Date, would be required to be listed in *Schedule 3.7(a)*,
- (iii) terminate (unless such Material Contract terminates pursuant to its stated terms) or materially amend the terms of any Material Contract;

(iv) settle any suit or litigation or waive any material claims or rights of material value in each case attributable to the Assets and effecting the period after the Effective Time;

- (v) transfer, sell, mortgage, pledge or dispose of any portion of the Assets other than the sale or disposal of Hydrocarbons in the ordinary course of business and sales of equipment that is no longer necessary in the operation of the Assets or for which replacement equipment of equal or greater value has been obtained; or
 - (vi) commit to do any of the foregoing.

26

Buyer acknowledges that SM Energy owns undivided interests in certain of the Assets with respect to which it is not the Operator, and Buyer agrees that the acts or omissions of the other Working Interests owners (including the Third Party Operators) who are not SM Energy shall not constitute a breach of the provisions of this Section 8.1, and no action required pursuant to a vote of Working Interest owners shall constitute a breach of the provisions of this Section 8.1 so long as SM Energy voted its interest in such a manner that complies with the provisions of this Section 8.1.

- **8.2** Governmental Bonds. Buyer acknowledges that none of the bonds, letters of credit and guarantees, if any, posted by SM Energy or its Affiliates with Governmental Authorities and relating to the Assets are transferable to Buyer. At or prior to Closing, Buyer shall deliver to SM Energy evidence of the posting of bonds or other security with all applicable Governmental Authorities meeting the requirements of such Governmental Authorities to own the Assets.
- 8.3 Notifications. If Buyer develops or possesses information that leads it to believe that SM Energy may have breached a representation, warranty, covenant or other agreement contained in this Agreement, Buyer shall promptly inform SM Energy of such potential breach so that SM Energy may attempt to remedy or cure such breach prior to Closing. The provisions of this Agreement relating to representations, warranties, indemnities and agreements of the Parties shall not be altered or modified by Buyer's knowledge or SM Energy's Knowledge, as applicable, of any event or Buyer's or SM Energy's review of any documents or other matters except as expressly provided herein to the contrary.
- 8.4 Suspense Accounts. SM Energy shall, prior to Closing, pay to the State of Texas or the State of Oklahoma, as applicable, those suspended royalties that, as of the Execution Date, have escheated to the State of Texas or the State of Oklahoma, as applicable. SM Energy shall transfer to Buyer via a downward adjustment to the Purchase Price only those monies held in suspense by SM Energy at Closing that relate to royalties as to which Buyer shall be responsible for distribution after Closing (which shall not include such escheated monies). SM Energy agrees to provide to Buyer any information reasonably requested by Buyer regarding all of SM Energy's accounts holding monies in suspense that will be transferred to Buyer together with a written explanation (as contained in SM Energy's files) of why such monies are held in suspense or other information identifying the proper disposition of such monies. Buyer agrees to take and apply such monies in a manner consistent with applicable Law and with prudent oil and gas business practices and to indemnify SM Energy against any claim relating to the failure to pay such funds after the Closing in accordance with the instructions or information delivered to Buyer by SM Energy.
- 8.5 Digital Records. After execution of this Agreement and within ten (10) Business Days prior to the Closing Date, SM Energy will deliver to Buyer, any requested Records available in digital format ("Digital Records") upon such request by Buyer so that Buyer may prepare to take over operations on the Closing Date. Any Digital Records delivered to Buyer prior to Closing will be subject to the public announcement provisions of Section 14.12 and the Confidentiality Agreement. If this Agreement is terminated prior to the Closing, then Buyer will return the Digital Records to SM Energy in their entirety and shall destroy any copies of the same that may have been created by Buyer.

8.6 Chieftain Litigation. At the end of each calendar quarter, Buyer shall invoice SM Energy for that portion of Reimbursable Chieftain Legal Expenses incurred or accrued in that quarter, and shall provide reasonable documentation and support for such invoice (the "Reimbursement Obligation"). SM Energy shall, within thirty (30) days of receipt of such invoice, pay the invoiced amount to Buyer at the account designated in the invoice. Any payment not made on or prior to the due date shall bear interest from the due date to the date of payment at the rate of interest published from time to time as the "Prime Rate" in the "Money Rates" section of The Wall Street Journal. Failure to include any reimbursable amount in a quarterly invoice shall not be deemed, or result in, a waiver of Buyer's right to be reimbursed for that amount.

ARTICLE IX CONDITIONS TO CLOSING

- **9.1** Buyer's Conditions to Closing. The obligations of Buyer to consummate the transactions contemplated by this Agreement are subject to the fulfillment (or waiver by Buyer) on or prior to the Closing of each of the following conditions:
- (a) Representations. The representations and warranties of SM Energy set forth in Article III shall be true and correct in all respects (other than those representations and warranties of SM Energy that are qualified by materiality, which shall be true and correct in all respects) as of the Closing Date as though made on and as of the Closing Date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except for those breaches, if any, of such representations and warranties that in the aggregate would not have a Material Adverse Effect.
- (b) <u>Performance</u>. SM Energy shall have materially performed or complied with all obligations, agreements and covenants contained in this Agreement as to which performance or compliance by SM Energy is required prior to or at the Closing Date.
- (c) No Legal Proceedings. No suit, action or other proceeding instituted by a third party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement. No order, award or judgment shall have been issued by any Governmental Authority or arbitrator to restrain, prohibit, enjoin, or declare illegal, or awarding substantial damages in connection with, the transactions contemplated by this Agreement.
- (d) <u>Title Defects; Environmental Defects; Preferential Rights; Consents.</u> The sum of (i) all (A) Title Defect Amounts determined under Section 6.2(c) (i), plus (B) all potential adjustments to the Purchase Price pursuant to Section 6.2(c) (ii) for Title Defects not cured as of the Closing, less (C) all Title Benefit Amounts determined under Section 6.2(g), plus (ii) all (A) Remediation Amounts for Environmental Defects determined under Section 7.1(b)(i), plus (B) all adjustments to the Purchase Price pursuant to Section 7.1(b)(ii), plus (iii) all adjustments to the Purchase Price made pursuant to Section 6.4(c)(ii) as a result of un-waived or unexpired Preferential Rights and Section 6.4(d)(i) in respect of unobtained or denied Consents, plus (iv) the amount of all Casualty Losses pursuant to Section 6.3 shall, in the aggregate, be less than 25% of the unadjusted aggregate Purchase Price.

28

- (e) <u>Closing Deliverables.</u> SM Energy shall have delivered (or be ready, willing and able to deliver at Closing) to Buyer the documents and other items required to be delivered by SM Energy under *Section 11.3*.
- **9.2 SM Energy's Conditions to Closing.** The obligations of SM Energy to consummate the transactions contemplated by this Agreement are subject to the fulfillment (or waiver by SM Energy) on or prior to the Closing of each of the following conditions:
- (a) Representations. The representations and warranties of Buyer set forth in Article IV shall be true and correct in all material respects (other than those representations and warranties of Buyer that are qualified by materiality, which shall be true and correct in all respects) as of the Closing Date as though made on and as of the Closing Date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date).
- (b) <u>Performance</u>. Buyer shall have materially performed or complied with all obligations, agreements and covenants contained in this Agreement as to which performance or compliance by Buyer is required prior to or at the Closing Date.
- (c) No Legal Proceedings. No suit, action or other proceeding instituted by a third party shall be pending before any Governmental Authority seeking to restrain, prohibit, enjoin or declare illegal, or seeking substantial damages in connection with, the transactions contemplated by this Agreement. No order, award or judgment shall have been issued by any Governmental Authority or arbitrator to restrain, prohibit, enjoin, or declare illegal, or awarding substantial damages in connection with, the transactions contemplated by this Agreement.
- (d) <u>Title Defects; Environmental Defects; Preferential Rights; Consents.</u> The sum of (i) all (A) Title Defect Amounts determined under Section 6.2(c) (i), plus (B) all potential adjustments to the Purchase Price pursuant to Section 6.2(c) (ii) for Title Defects not cured as of the Closing, less (C) all Title Benefit Amounts determined under Section 6.2(g), plus (ii) all (A) Remediation Amounts for Environmental Defects determined under Section 7.1(b)(i), plus (B) all adjustments to the Purchase Price pursuant to Section 7.1(b)(ii), plus (iii) all adjustments to the Purchase Price made pursuant to Section 6.4(c)(ii) as a result of un-waived or unexpired Preferential Rights and Section 6.4(d)(i) in respect of unobtained or denied Consents, plus (iv) the amount of all Casualty Losses pursuant to Section 6.3 shall, in the aggregate, be less than 25% of the unadjusted aggregate Purchase Price.
- (e) <u>Closing Deliverables.</u> Buyer shall have delivered (or be ready, willing and able to deliver at Closing) to SM Energy the documents and other items required to be delivered by Buyer under Section 11.3.

ARTICLE X TAX MATTERS

10.1 Asset Tax Liability. Subject to the treatment of ad valorem Taxes provided below, all Asset Taxes shall be allocated between Buyer and SM Energy as of the Effective Time for all taxable periods that include the Effective Time. All Asset Taxes that are not ad valorem

29

taxes shall be allocated to SM Energy to the extent they relate to production prior to the Effective Time and to Buyer to the extent they relate to production on or after the Effective Time. No liability for Asset Taxes shall duplicate an adjustment to Purchase Price made pursuant to Section 2.4. Ad valorem Taxes for each assessment period shall be allocated to SM Energy based on the percentage of the assessment period occurring before the Effective Time and to Buyer based on the percentage of the assessment period occurring on or after the Effective Time. Each Party shall promptly furnish to the other copies of any Asset Tax assessments and statements (or invoices therefor from the operator of the Assets) received by it to the extent such assessment, statement, or invoice relates to an Asset Tax allocable to the other Party under this Section 10.1. Each Party shall timely pay all Asset Taxes subject to allocation under this Section and shall furnish to the other Party evidence of such payment. The Parties shall estimate all Asset Taxes asserted against it that are attributable to the ownership or operation of the Assets to the extent they relate to the period on and after the Effective Time and

through the Closing Date and all Subject Transfer Taxes and incorporate such estimates into the Preliminary Settlement Statement. The actual amounts (to the extent the actual amounts differ from the estimates included in the Preliminary Settlement Statement and are known at the time of the Final Settlement Statement) shall be accounted for in the Final Settlement Statement. If the actual amounts are not known at the time of the Final Settlement, the amounts shall be re-estimated based on the best information available at the time of the Final Settlement. When the actual amounts are known, Buyer or SM Energy shall make such payments to the other (if any) as are necessary to effect the allocation of Taxes described in this Section 10.1.

- 10.2 Transfer Taxes. All sales, use or other Taxes (other than Taxes on gross income, net income, gross receipts or capital gains, or franchise taxes), duties, levies, recording fees or other governmental charges incurred by or imposed with respect to the property transfers undertaken pursuant to this Agreement ("Subject Transfer Taxes") shall be the responsibility of, and shall be paid by, Buyer. The Parties shall reasonably cooperate in taking steps that would minimize or eliminate any Subject Transfer Taxes. Buyer agrees to file all Subject Transfer Tax Returns relating to such Subject Transfer Taxes.
- 10.3 Asset Tax Reports and Returns. For Asset Tax periods in which the Effective Time occurs, SM Energy agrees to immediately forward to Buyer copies of any Asset Tax reports and Tax Returns received or filed by SM Energy after Closing and provide Buyer with any information SM Energy has that is reasonably necessary for Buyer to file any required Tax Return related to the Assets. Buyer agrees to file all Tax Returns and reports applicable to the Assets that Buyer is required to file after the Closing and, subject to the provisions of Section 10.1, to pay all required Asset Taxes payable with respect to the Assets.
- 10.4 Tax Cooperation. Buyer and SM Energy shall cooperate fully as and to the extent reasonably requested by the other party, in connection with the filing of any Tax Returns and any audit, litigation or other proceeding (each, a "Tax Proceeding") with respect to Taxes relating to or in connection with the Assets. Such cooperation shall include the retention and (upon the other Party's request) the provision of such records and information which are reasonably relevant to any such Tax Return or Tax Proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.

30

ARTICLE XI CLOSING

- 11.1 Date of Closing. Subject to the conditions stated in this Agreement, the transfer by SM Energy and the acceptance by Buyer of the Assets (the 'Closing') shall occur on December 30, 2013 or, if all conditions to Closing in Article IX (other than those conditions that are only capable of being satisfied at the Closing) have not yet been satisfied or waived by that date, five (5) Business Days after such conditions have been satisfied or waived, or such other date as Buyer and SM Energy may agree upon in writing. The date when Closing actually occurs shall be the "Closing Date."
- 11.2 Place of Closing. Closing shall be held at the offices of SM Energy, at 1775 Sherman Street, Suite 1200, Denver, CO 80203 or such other location as Buyer and SM Energy may agree upon in writing.
- 11.3 Closing Obligations. At Closing, the following documents shall be delivered and the following events shall occur, the execution of each document and the occurrence of each event being a condition precedent to the others and each being deemed to have occurred simultaneously with the others:
- (a) SM Energy and Buyer shall execute and deliver the Assignment, in sufficient counterparts to facilitate recording in the applicable counties where the Assets are located:
- (b) SM Energy and Buyer shall execute and deliver assignments, on appropriate forms, of state and of federal leases comprising portions of the Assets, if any;
 - (c) SM Energy and Buyer shall execute and deliver the Preliminary Settlement Statement pursuant to Section 2.6(a);
- (d) Buyer shall deliver to SM Energy, to the accounts designated in the Preliminary Settlement Statement, by direct bank or wire transfer in same day funds, the Closing Amount;
- (e) SM Energy and Buyer shall, in accordance with the terms of the Escrow Agreement, instruct the Escrow Agent to distribute the Deposit in same day funds to the account designated by SM Energy;
- (f) SM Energy shall deliver on forms supplied by Buyer (and reasonably acceptable to SM Energy) transfer orders or letters in lieu thereof directing all purchasers of production to make payment to Buyer of proceeds attributable to Hydrocarbon production from the Assets from and after the Effective Time, for delivery by Buyer to each purchaser of such Hydrocarbon production;
- (g) SM Energy shall deliver an executed statement described in Treasury Regulation § 1.1445-2(b)(2) certifying that SM Energy is not a "foreign person" or a "disregarded entity";

31

- (h) Buyer shall execute and deliver a certificate from an authorized officer of Buyer certifying on behalf of Buyer that the conditions set forth in Section 9.2(a) and Section 9.2(b) have been fulfilled by Buyer;
- (i) SM Energy shall execute and deliver a certificate from an authorized officer of SM Energy certifying on behalf of SM Energy that the conditions set forth in Section 9.1(a) and Section 9.1(b) have been fulfilled by SM Energy;
- (j) SM Energy shall deliver a recordable release of any trust, mortgages, financing statements, fixture filings and security agreements made by SM Energy or its Affiliates affecting the Assets; and
- (k) SM Energy and Buyer shall execute and deliver any other agreements, instruments and documents that are required by other terms of this Agreement to be executed or delivered at Closing.
- 11.4 Records. In addition to the obligations set forth under Section 11.3 above, as soon as reasonably practicable following Closing but in any event within 10 days following the Closing Date, SM Energy shall make available to Buyer, during normal business hours at SM Energy's offices, the Records.

ARTICLE XII ACQUISITION TERMINATION AND REMEDIES

- (a) by SM Energy, if any of the conditions set forth in Section 9.2 (other than the conditions set forth in Section 9.2(d)) have not been satisfied by Buyer on or before January 31, 2014 (the "Outside Termination Date");
- (b) by Buyer, if any of the conditions set forth in Section 9.1 (other than the conditions set forth in Section 9.1(d)) have not been satisfied by SM Energy on or before the Outside Termination Date;
- (c) by SM Energy if the condition set forth in Section 9.2(d) is not satisfied on or before the Closing Date or by Buyer if the condition set forth in Section 9.1(d) has not been satisfied on or before the Closing Date; or
 - (d) by the mutual written agreement of Buyer and SM Energy;

provided, however, that no Party shall have the right to terminate this Agreement pursuant to clause (a) or (b) above if such Party or its Affiliates are at such time in material breach of any provision of this Agreement.

12.2 Effect of Termination. If the obligation to close the transactions contemplated by this Agreement is terminated pursuant to any provision of Section 12.1, then, except for the provisions of (a) Section 2.3, Sections 5.1(c) through 5.1(f), Section 5.2, Section 5.3, this

32

Section 12.2 and Section 13.9, and (b) such terms as set forth in this Agreement in order to give context to any of the surviving Sections, this Agreement shall forthwith become void and the Parties shall have no liability or obligation hereunder except and to the extent such termination results from the willful breach by a Party of any of its covenants or agreements hereunder, in which case the non-breaching Party shall be entitled to all remedies available at Law or in equity, including specific performance, and shall be entitled to recover court costs and reasonable attorneys' fees in addition to any other relief to which such Party or Parties may be entitled.

- 12.3 Specific Performance. The Parties agree that if any of the provisions of this Agreement were not performed in accordance with their specific terms, irreparable damage would occur, no adequate remedy at law would exist and damages would be difficult to determine, and the Parties shall be entitled to specific performance of the terms hereof and immediate injunctive relief, without the necessity of proving the inadequacy of money damages as a remedy, in addition to any other remedy available at law or in equity.
- 12.4 Return of Documentation and Confidentiality. Upon termination of this Agreement, Buyer shall return to SM Energy all title, engineering, geological and geophysical data, environmental assessments or reports, maps and other information furnished by SM Energy to Buyer or, if not destroyed by Buyer, prepared by or on behalf of Buyer in connection with its due diligence investigation of the Assets, in each case, in accordance with the Confidentiality Agreement, and an officer of Buyer shall certify same to SM Energy in writing.

ARTICLE XIII ASSUMPTION; SURVIVAL; INDEMNIFICATION

13.1 Assumption by Buyer. Without limiting Buyer's rights to indemnity under this Article XIII and subject to any adjustments to the Purchase Price pursuant to Section 2.4, from and after Closing, Buyer assumes and agrees to fulfill, perform, pay and discharge (or cause to be fulfilled, performed, paid or discharged) all obligations and Liabilities, known or unknown, with respect to the Assets, regardless of whether such obligations or Liabilities arose prior to, on or after the Effective Time, including, but not limited to, obligations and Liabilities relating in any manner to the use, ownership or operation of the Assets, such as obligations to: (a) furnish makeup gas and/or settle Imbalances attributable to the Assets according to the terms of applicable gas sales, processing, gathering or transportation Contracts, (b) pay working interests, royalties, overriding royalties and other interest owners' revenues or proceeds attributable to sales of Hydrocarbons produced from the Assets, including those held in suspense, (c) pay the proportionate share attributable to the Assets to properly gand abandon any and all wells, including inactive wells or temporarily abandoned wells, located on the Assets, (d) pay the proportionate share attributable to the Assets to replug any well, wellbore or previously plugged well on the Assets to the extent required or necessary, (e) pay the proportionate share attributable to the Assets to dismantle or decommission and remove any Personal Property and other property of whatever kind related to or associated with operations and activities conducted by whomever on the Assets, (f) pay the proportionate share attributable to the Assets to clean up, restore and/or remediate the Assets in accordance with Applicable Contracts and Laws, and (g) pay the proportionate share attributable to the Assets to perform all obligations applicable to or imposed on the lessee, owner or operator under the Leases and the Applicable Contracts, or as required by any Law, including the payment of all Tax

33

obligations and Liabilities, subject to the exclusions below, herein being referred to as the "Assumed Obligations"); provided, Buyer does not assume any obligations or Liabilities of SM Energy attributable to the Assets to the extent that such obligations or Liabilities consist of any of the following (the "Retained Obligations"):

- (i) attributable to or arising out of the ownership, use or operation of the Excluded Assets;
- (ii) attributable to any Income Tax Liability or Franchise Tax Liability; or
- (iii) the Chieftain Covered Legal Expenses and the Reimbursement Obligation,

all of which Retained Obligations SM Energy retains and SM Energy agrees to fulfill, perform, pay and discharge (or cause to be fulfilled, performed, paid or discharged) all obligations and liabilities, known or unknown, with respect to the Retained Obligations.

- 13.2 Indemnities of SM Energy. Effective as of the Closing, subject to the limitations set forth in Section 13.4 and otherwise contained in this Article XIII, SM Energy is responsible for, shall pay on a current basis and agrees to defend, indemnify and hold harmless Buyer and its Affiliates, and all of its and their respective stockholders, partners, members, directors, officers, managers, employees, agents and representatives (collectively, "Buyer Indemnified Parties") from and against any and all Liabilities, arising from, based upon, related to or associated with:
 - (a) any breach by SM Energy of its representations or warranties contained in Article III;
 - (b) any breach by SM Energy of its covenants and agreements contained in this Agreement;
 - (c) the Retained Obligations;
 - (d) costs and expenses for goods and services attributable to SM Energy's ownership of the Assets prior to the Effective Time;
 - (e) Asset Taxes attributable to SM Energy's ownership of the Assets prior to the Effective Time;
 - (f) Liabilities to third parties for personal injury or death attributable to SM Energy's operation of the Assets prior to the Effective Time; or

- (g) Liabilities to third parties for disposal or transportation of Hazardous Substances from the Assets attributable to SM Energy's operation of the Assets prior to the Effective Time.
- 13.3 Indemnities of Buyer. Effective as of the Closing, Buyer and their respective successors and assigns shall assume, be responsible for, shall pay on a current basis and agree to

34

defend, indemnify, hold harmless and forever release SM Energy and its Affiliates, and all of their respective stockholders, partners, members, directors, officers, managers, employees, agents and representatives (collectively, "SM Indemnified Parties") from and against any and all Liabilities arising from, based upon, related to or associated with:

- (a) any breach by Buyer of its representations or warranties contained in *Article IV*;
- (b) any breach by Buyer of its covenants and agreements contained in this Agreement; or
- (c) the Assumed Obligations.

13.4 Limitation on Liability.

- (a) SM Energy shall not have any liability for any indemnification under Section 13.2 unless (i) the individual amount of any Liability for which a Claim Notice is delivered by Buyer to SM Energy under this Article XIII and for which SM Energy is liable exceeds [***] and (ii) except for breaches of the representations and warranties in Sections 3.1, 3.2, 3.3, 3.4 and 3.5 and except for the Retained Obligations (to which the deductible shall not apply), the aggregate amount of such Liabilities for which SM Energy is liable under this Agreement after the application of the provisions of clause (i) above exceeds [***] of the aggregate unadjusted Purchase Price.
- (b) For purposes of this *Article XIII*, any breach or inaccuracy in any representations or warranties shall be determined without regard to any dollar or materiality qualifiers.
- (c) Notwithstanding anything to the contrary contained in this Agreement, SM Energy shall not be required to indemnify the Buyer Indemnified Parties for aggregate Liabilities under *Section 13.2* in excess of [***] of the aggregate unadjusted Purchase Price, except for breaches of the representations and warranties in *Sections 3.1, 3.2, 3.3, 3.4* and *3.5* and except for the Retained Obligations to which this limitation shall not apply.
- 13.5 Express Negligence. EXCEPT AS OTHERWISE PROVIDED IN SECTION 5.1(c), THE INDEMNIFICATION, RELEASE, ASSUMED OBLIGATIONS, WAIVER AND LIMITATION OF LIABILITY PROVISIONS PROVIDED FOR IN THIS AGREEMENT SHALL BE APPLICABLE WHETHER OR NOT THE LIABILITIES, LOSSES, COSTS, EXPENSES AND DAMAGES IN QUESTION AROSE OR RESULTED SOLELY OR IN PART FROM THE GROSS, SOLE, ACTIVE, PASSIVE, CONCURRENT OR COMPARATIVE NEGLIGENCE, STRICT LIABILITY OR OTHER FAULT OR VIOLATION OF LAW OF OR BY ANY INDEMNIFIED PERSON. BUYER AND SM ENERGY ACKNOWLEDGE THAT THIS STATEMENT COMPLIES WITH THE EXPRESS NEGLIGENCE RULE AND IS "CONSPICUOUS".
- 13.6 Exclusive Remedy for Agreement. Notwithstanding anything to the contrary contained in this Agreement, from and after Closing, Section 5.1(c), Section 13.2 and Section 13.3 contain the Parties' exclusive remedy against each other with respect to breaches of

35

the representations, warranties, covenants and agreements of the Parties contained in this Agreement and the affirmations of such representations, warranties, covenants and agreements contained in the certificate delivered by each Party at Closing pursuant to Section 11.3(h) or Section 11.3(i), as applicable. Except for (a) the remedies contained in this Article XIII, and (b) other remedies available to the Parties at Law or in equity for breaches of Section 5.1(c) and Section 5.2, from and after Closing, SM Energy and Buyer each release, remise and forever discharge the other Party and its Affiliates and all such Persons' stockholders, officers, directors, employees, agents, advisors and representatives from any and all Liabilities in Law or in equity, known or unknown, which such Parties might now or subsequently may have, based on, relating to or arising out of (i) this Agreement or the consummation of the transactions contemplated by this Agreement, (ii) the ownership, use or operation of the Assets prior to the Closing, or the condition, quality, status or nature of the Assets prior to the Closing, including rights to contribution under the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended, (iii) breaches of statutory or implied warranties with respect to this Agreement, (iv) nuisance or other tort actions with respect to this Agreement, (v) rights to punitive damages with respect to this Agreement, (vi) common Law rights of contribution with respect to this Agreement, and (vii) rights under insurance maintained by SM Energy or any of its Affiliates with respect to this Agreement.

- 13.7 Indemnification Procedures. All claims for indemnification under Section 5.1(c), Section 13.2 and Section 13.3 shall be asserted and resolved as follows:
- (a) For purposes of this *Article XIII*, the term "*Indemnifying Party*", when used in connection with particular Liabilities, shall mean the Party having an obligation to indemnify another Party or Person(s) with respect to such Liabilities pursuant to this *Article XIII*, and the term "*Indemnified Party*", when used in connection with particular Liabilities, shall mean the Party or Person(s) having the right to be indemnified with respect to such Liabilities by another Party pursuant to this *Article XIII*.
- (b) To make claim for indemnification under Section 5.1(c), Section 13.2 or Section 13.3, an Indemnified Party shall notify the Indemnifying Party of its claim under this Section 13.7, including the specific details of and specific basis under this Agreement for its claim (the 'Claim Notice'). In the event that the claim for indemnification is based upon a claim by a third party against the Indemnified Party (a "Third Party Claim"), the Indemnified Party shall provide its Claim Notice promptly after the Indemnified Party has actual knowledge of the Third Party Claim and shall enclose a copy of all papers (if any) served with respect to the Third Party Claim; provided that the failure of any Indemnified Party to give notice of a Third Party Claim as provided in this Section 13.7 shall not relieve the Indemnifying Party of its obligations under Section 5.1(c), Section 13.2 or Section 13.3 (as applicable) except to the extent such failure results in insufficient time being available to permit the Indemnifying Party to effectively defend against the Third Party Claim or otherwise materially prejudices the Indemnifying Party's ability to defend against the claim. In the event that the claim for indemnification is based upon an inaccuracy or breach of a representation, warranty, covenant or agreement, the Claim Notice shall specify the representation, warranty, covenant or agreement that was inaccurate or breached.

36

other pleading that it shall deem necessary or appropriate to protect its interests or those of the Indemnifying Party and that is not prejudicial to the Indemnifying Party.

- (d) If the Indemnifying Party admits its obligation to indemnify a Third Party Claim, it shall have the right and obligation to diligently defend, at its sole cost and expense, the Third Party Claim provided that, where the Third Party Claim consists of a civil, criminal or regulatory proceeding, action, indictment or investigation against the Indemnified Party by any Governmental Authority, the Indemnified Party shall at its option have the right to control the defense and proceedings. Except as provided in the preceding sentence, the Indemnifying Party shall have full control of such defense and proceedings, including any compromise or settlement thereof. If requested by the Indemnifying Party, the Indemnified Party agrees to cooperate in contesting any Third Party Claim that the Indemnifying Party elects to contest (provided, however, that the Indemnified Party shall not be required to bring any counterclaim or cross-complaint against any Person). The Indemnified Party may participate in, but not control, at its own expense, any defense or settlement of any Third Party Claim controlled by the Indemnifying Party pursuant to this Section 13.7(d). An Indemnifying Party shall not, without the written consent of the Indemnified Party, (i) settle any Third Party Claim or consent to the entry of any judgment with respect thereto which does not result in a final resolution of the Indemnified Party's Liability in respect of such Third Party Claim (including in the case of a settlement an unconditional written release of the Indemnified Party from all Liability in respect of such Third Party Claim) or (ii) settle any Third Party Claim or consent to the entry of any judgment with respect thereto in any manner that may materially and adversely affect the Indemnified Party (other than as a result of money damages covered by the indemnity).
- (e) If the Indemnifying Party does not admit its obligation to indemnify and bear all expenses associated with a Third Party Claim or admits its obligation to indemnify and bear all expenses associated with a Third Party Claim but fails to diligently prosecute or settle the Third Party Claim, then the Indemnified Party shall have the right to defend against the Third Party Claim at the sole cost and expense of the Indemnifying Party, with counsel of the Indemnified Party's choosing, subject to the right of the Indemnifying Party to admit its obligation to indemnify and bear all expenses associated with a Third Party Claim and assume the defense of the Third Party Claim at any time prior to settlement or final determination thereof. If the Indemnifying Party has not yet admitted its obligation to indemnify and bear all expenses associated with a Third Party Claim, the Indemnified Party shall send written notice to the Indemnifying Party of any proposed settlement and the Indemnifying Party shall have the option for 10 Business Days following receipt of such notice to (i) admit in writing its obligation to indemnify and bear all expenses associated with a Third Party Claim and (ii) if such obligation is so admitted, reject, in its reasonable judgment, the proposed settlement.
- (f) In the case of a claim for indemnification not based upon a Third Party Claim, the Indemnifying Party shall have 30 days from its receipt of the Claim Notice to (i) cure

37

the Liabilities complained of, (ii) admit its obligation to indemnify for and bear all expenses associated with such Liability or (iii) dispute the claim for such Liabilities. If the Indemnifying Party does not notify the Indemnified Party within such 30 day period that it has cured the Liabilities or that it disputes the claim for such Liabilities, the amount of such Liabilities shall conclusively be deemed a liability of the Indemnifying Party hereunder.

13.8 Survival.

- (a) The representations and warranties of SM Energy contained in Sections 3.1, 3.2, 3.3, 3.4 and 3.5 shall, in each case, survive the Closing indefinitely. The representations and warranties of SM Energy contained in Sections 3.6 through 3.18 shall survive the Closing for a period of twelve (12) Months. The representations and warranties of Buyer contained in Article IV shall survive the Closing indefinitely. Subject to Section 13.8(b), the covenants and agreements of SM Energy (including the Retained Obligations) and Buyer contained in this Agreement shall, in each case, survive indefinitely. Representations, warranties, covenants and agreements shall be of no further force and effect after the date of their expiration; provided that there shall be no termination of any bona fide claim asserted pursuant to this Agreement with respect to such a representation, warranty, covenant or agreement prior to its expiration date.
- (b) The indemnities in Section 13.2(a) and Section 13.2(b) shall terminate as of the termination date of each respective representation, warranty, covenant or agreement that is subject to indemnification as set forth in Section 13.8(a). The indemnities in Sections 13.2(d), (e), (f) and (g) shall survive Closing for twelve (12) Months. Buyer's indemnities contained in Section 13.3 and SM Energy's indemnity in Section 13.2(c) shall survive Closing without time limit. Notwithstanding the foregoing, there shall be no termination of any bona fide claim asserted pursuant to the indemnities in Section 13.2 prior to the date of termination for such indemnity.
- 13.9 Non-Compensatory Damages. None of the Buyer Indemnified Parties or SM Indemnified Parties shall be entitled to recover from SM Energy or Buyer, as applicable, or their respective Affiliates, any indirect, consequential, punitive or exemplary damages or damages for lost profits of any kind arising under or in connection with this Agreement or the transactions contemplated by this Agreement, except to the extent any such Party suffers such damages (including costs of defense and reasonable attorney's fees incurred in connection with defending of such damages) to a third party, which damages (including costs of defense and reasonable attorney's fees incurred in connection with defending against such damages) shall not be excluded by this provision as to recovery hereunder. Subject to the preceding sentence, Buyer, on behalf of each of the Buyer Indemnified Parties, and SM Energy, on behalf of each of SM Indemnified Parties, each waive any right to recover punitive, special, exemplary and consequential damages, including damages for lost profits of any kind, arising in connection with this Agreement or the transactions contemplated by this Agreement. This Section 13.9 shall not restrict any Party's right to obtain specific performance or other equitable remedies (other than rescission) pursuant to Section 12.2.

38

ARTICLE XIV MISCELLANEOUS

- 14.1 Counterparts. This Agreement may be executed in any number of counterparts, and each such counterpart hereof shall be deemed to be an original instrument, but all of such counterparts shall constitute for all purposes one agreement. Any signature hereto delivered by a Party by facsimile transmission or by email delivery of a ".pdf" format data file shall be deemed an original signature hereto.
- 14.2 Notices. All notices and communications required or permitted to be given hereunder shall be sufficient in all respects if given in writing and delivered personally, or sent by bonded overnight courier, or mailed by U.S. Express Mail or by certified or registered United States Mail with all postage fully prepaid, or sent by facsimile transmission or electronic mail (provided any such facsimile transmission or electronic mail is confirmed either orally or by written confirmation, and provided further that any notice of any breach or default of this Agreement may not be delivered by electronic mail), addressed to the appropriate Party at the address for such Party shown below or at such other address as such Party shall have theretofore designated by written notice delivered to the Party giving such notice:

If to SM Energy:

SM Energy Company 1775 Sherman Street, Suite 1200 Denver, CO 80203

Attention: Herbert S. Vogel — Senior Vice President — Portfolio Development and Technical Services

Fax: 303.864.2598

Email: hvogel@sm-energy.com

With a copy, which shall not constitute notice to:

SM Energy Company 1775 Sherman Street, Suite 1200

Denver, CO 80203

Attention: David W. Copeland — Executive Vice President, General Counsel and Corporate Secretary

Fax: 303.864.2598

Email: dcopeland@sm-energy.com

39

If to Buyer:

Enervest Energy Institutional Fund XIII-A, L.P. Enervest Energy Institutional Fund XIII-WIB, L.P. Enervest Energy Institutional Fund XIII-WIC, L.P. c/o EnerVest, Ltd. 1001 Fannin, Suite 800 Houston, Texas 77002

Attention: Fabené J. Welch Facsimile: (713) 982-1500 Email: fwelch@enervest.net

With a copy, which shall not constitute notice to:

c/o EnerVest, Ltd. 1001 Fannin, Suite 800 Houston, Texas 77002

Attention: Philip B. Berry
Facsimile: (713) 615-7799
Email: pberry@enervest.net

Any notice given in accordance herewith shall be deemed to have been given when (a) delivered to the addressee in person or by courier, (b) transmitted by facsimile transmission during normal business hours with written receipt confirmation, (c) sent by electronic mail during normal business hours, or (d) upon actual receipt by the addressee after such notice has either been delivered to an overnight courier or deposited in the United States Mail, as the case may be. The Parties may change the address and facsimile numbers and electronic mail address to which such communications are to be addressed by giving written notice to the other Parties in the manner provided in this Section 14.2.

- 14.3 Expenses. Except as otherwise specifically provided, all fees, costs and expenses incurred by the Parties in negotiating this Agreement shall be paid by the Party incurring the same, including legal and accounting fees, costs and expenses.
- 14.4 Waivers; Rights Cumulative. Any of the terms, covenants or conditions hereof may be waived only by a written instrument executed by or on behalf of the Party waiving compliance. No course of dealing on the part of any Party or its respective officers, employees, agents or representatives, and no failure by a Party to exercise any of its rights under this Agreement shall, in either case, operate as a waiver thereof or affect in any way the right of such Party at a later time to enforce the performance of such provision. No waiver by any Party of any condition, or any breach of any term or covenant contained in this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such condition or breach or a waiver of any other condition or of any breach of any other term or covenant. The rights of the Parties under this Agreement shall be cumulative, and the exercise or partial exercise of any such right shall not preclude the exercise of any other right.

40

- 14.5 Relationship of the Parties. The rights, duties, obligations and liabilities of the Parties under this Agreement shall be individual, not joint or collective. It is not the intention of the Parties to create, and this Agreement shall not be deemed or construed to create, a mining or other partnership, joint venture or association or a trust. This Agreement shall not be deemed or construed to authorize any Party to act as an agent, servant or employee for any other Party for any purpose whatsoever except as explicitly set forth in this Agreement. In their relations with each other under this Agreement, the Parties shall not be considered fiduciaries.
- 14.6 Entire Agreement; Conflicts. THIS AGREEMENT, THE EXHIBITS, SCHEDULES AND APPENDICES HERETO COLLECTIVELY CONSTITUTE THE ENTIRE AGREEMENT AMONG THE PARTIES PERTAINING TO THE SUBJECT MATTER HEREOF AND SUPERSEDE ALL PRIOR AGREEMENTS, UNDERSTANDINGS, NEGOTIATIONS AND DISCUSSIONS, WHETHER ORAL OR WRITTEN, OF THE PARTIES PERTAINING TO THE SUBJECT MATTER OF THIS AGREEMENT. THERE ARE NO WARRANTIES, REPRESENTATIONS OR OTHER AGREEMENTS AMONG THE PARTIES RELATING TO THE SUBJECT MATTER OF THIS AGREEMENT EXCEPT AS SPECIFICALLY SET FORTH IN THIS AGREEMENT, AND NO PARTY SHALL BE BOUND BY OR LIABLE FOR ANY ALLEGED REPRESENTATION, PROMISE, INDUCEMENT OR STATEMENTS OF INTENTION NOT SO SET FORTH. IN THE EVENT OF A CONFLICT BETWEEN THE TERMS AND PROVISIONS OF THIS AGREEMENT AND THE TERMS AND PROVISIONS OF ANY EXHIBIT HERETO; THE TERMS AND PROVISIONS OF THIS AGREEMENT SHALL GOVERN AND CONTROL; PROVIDED, HOWEVER, THAT THE INCLUSION IN ANY OF THE EXHIBITS HERETO OF TERMS AND PROVISIONS NOT ADDRESSED IN THIS AGREEMENT SHALL NOT BE DEEMED A CONFLICT, AND ALL SUCH ADDITIONAL PROVISIONS SHALL BE GIVEN FULL FORCE AND EFFECT.
- 14.7 Governing Law. THIS AGREEMENT, THE OTHER DOCUMENTS DELIVERED PURSUANT HERETO, AND THE LEGAL RELATIONS BETWEEN THE PARTIES SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS, EXCLUDING ANY CONFLICTS OF LAW RULE OR PRINCIPLE THAT MIGHT REFER TO THE LAWS OF ANOTHER JURISDICTION. THE VALIDITY OF THE VARIOUS CONVEYANCES AFFECTING THE TITLE TO REAL PROPERTY IN OKLAHOMA SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF OKLAHOMA. ANY DISPUTE SHALL BE BROUGHT BEFORE THE FEDERAL OR STATE COURTS SITTING IN THE CITY OF HOUSTON AND COUNTY OF HARRIS, TEXAS, AND THE PARTIES IRREVOCABLY CONSENT TO THE EXCLUSIVE JURISDICTION OF SUCH COURTS AND WAIVE ANY OBLIGATION TO THE VENUE (INCLUDING ON THE BASIS OF INCONVENIENT FORUM) AND WAIVE ANY RIGHT TO CHOOSE OR REQUEST ANY OTHER VENUE. EACH OF THE PARTIES CONSENTS TO THE SERVICE OF PROCESS IN ANY SUIT, ACTION, OR PROCEEDING IN ANY OF THE AFORESAID COURTS BY THE SERVICE OF PROCESS ON ANY PARTY ANYWHERE IN THE WORLD, INCLUDING BY MAILING COPIES OF PROCESS TO SUCH PARTY BY CERTIFIED OR REGISTERED MAIL TO THE ADDRESS SET FORTH IN SECTION 14.2. NOTHING IN THIS SECTION 14.7 WILL AFFECT THE RIGHT OF ANY PARTY TO SERVE LEGAL PROCESS IN ANY OTHER MANNER PERMITTED BY LAW OR AT EQUITY. EACH

PARTY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY DISPUTE.

- 14.8 Filings, Notices and Certain Governmental Approvals. Promptly after Closing, Buyer shall (a) record the Assignments and all state/federal assignments executed at Closing in all applicable real property records and/or, if applicable, all state or federal agencies, (b) send notices to vendors supplying goods and services for the Assets of the assignment of the Assets to Buyer, (c) actively pursue the approval of all applicable Governmental Authorities of the assignment of the Assets to Buyer and (d) actively pursue all other consents and approvals that may be required in connection with the (i) assignment of the Assets to Buyer and (ii) assumption of the Assumed Liabilities by Buyer hereunder, in each case, that shall not have been obtained prior to Closing. Buyer obligates itself to take any and all action required by any Governmental Authority in order to obtain such unconditional approval, including but not limited to, the posting of any and all bonds or other security that may be required in excess of any existing bond.
- 14.9 Amendment. This Agreement may be amended only by an instrument in writing executed by all of the Parties and expressly identified as an amendment or modification hereof.
 - 14.10 Parties in Interest. Nothing in this Agreement shall entitle any Person other than the Parties to any claim, cause of action, remedy or right of any kind.
- 14.11 Successors and Permitted Assigns. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns.

14.12 Publicity.

- (a) Without reasonable prior notice and consent from the other Parties, no Party will issue, or permit any of its agents or Affiliates to issue, any press releases or otherwise make, or cause any of its agents or Affiliates to make, any public statements with respect to this Agreement or the activities contemplated hereby, except where such release or statement is deemed in good faith by the releasing Party to be required by Law or under the rules and regulations of a recognized stock exchange on which shares of such Party or any of its Affiliates are listed.
- (b) Notwithstanding anything to the contrary in Section 14.12(a), any Party or Affiliate of a Party may disclose information regarding the Assets in investor presentations, in industry conference presentations or similar disclosures, so long as, if in industry conference presentations or similar disclosures, the other Party is not identified by name.
- 14.13 Preparation of Agreement. Both SM Energy and Buyer and their respective counsel participated in the preparation of this Agreement. In construing this Agreement, no presumption shall arise based on the identity of the draftsman of this Agreement.
- 14.14 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any adverse manner to any Party. Upon such determination that any term or other provision is

42

invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the extent possible.

[Signature Page Follows]

43

IN WITNESS WHEREOF, the Parties have executed this Agreement by their duly authorized representatives on and as of the Execution Date.

SM ENERGY COMPANY

By: /s/ Javan D. Ottoson

Name: Javan D. Ottoson

Title: President and Chief Operating Officer

ENERVEST ENERGY INSTITUTIONAL FUND XIII—A, L.P. ENERVEST ENERGY INSTITUTIONAL FUND XIII—WIB, L.P.

By: EnerVest, Ltd.,

its Managing General Partner

By: EnerVest Management GP, L.C.,

its General Partner

By: /s/ John B. Walker

Name: John B. Walker

Title: President and Chief Executive Officer

ENERVEST ENERGY INSTITUTIONAL FUND XIII—WIC, L.P.

By: EnerVest Holding XIII, LLC,

its General Partner

By: EnerVest, Ltd.,

its Sole Member

By: EnerVest Management GP, L.C.,

its General Partner

By: /s/ John B. Walker

Name: John B. Walker

Title: President and Chief Executive Officer

SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT

Annex I

Definitions

"Accounting Arbitrator" has the meaning set forth in Section 2.6(c).

"AFE" has the meaning set forth in Section 3.11.

"Affiliate" means, with respect to any Party, a Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by or is under common Control with, such Party.

"Aggregate Deductible" has the meaning set forth in Section 6.2(h).

"Agreement" has the meaning set forth in the preamble to this Agreement.

"Allocated Value" has the meaning set forth in Section 2.5.

"Applicable Contracts" means all Contracts to which SM Energy is a party that primarily relate to the Assets and that will be binding on the Assets or Buyer after Closing, including, without limitation; farmin and farmout agreements; bottomhole agreements; crude oil, condensate and natural gas purchase and sale agreements; gathering, transportation and marketing agreements; hydrocarbon storage agreements; acreage contribution agreements; operating agreements (including, for the avoidance of doubt, Applicable Operating Agreements); balancing agreements; pooling declarations or agreements; unitization agreements; processing agreements and other similar contracts and agreements, but excluding the Leases and any master service agreements or other agreements held by SM Energy in its capacity as Operator of the Assets.

"Applicable Operating Agreements" means, collectively, the joint operating agreements applicable to the Assets, and "Applicable Operating Agreement" means any of them.

"Asset Taxes" means ad valorem, property, excise, severance, production or similar taxes (including any interest, fine, penalty or additions to tax imposed by Governmental Authorities in connection with such taxes) based upon operation or ownership of the Assets or the production of Hydrocarbons therefrom, but excluding, for the avoidance of doubt, income, capital gains and franchise taxes.

"Assets" means SM Energy's right, title and interest in and to the following: (a) all oil and gas leases and mineral interests described on Exhibit A and any leasehold estates, royalty interests, overriding royalty interests, net profits interests, and other rights and interests to the oil and gas in place covered by such leases (the "Leases") and any pooled acreage, communitized acreage or units arising on account of Leases being pooled, communitized or unitized into such units ("Units"); (b) the oil, gas, casinghead gas, coal bed methane, condensate and other gaseous and liquid hydrocarbons or any combination thereof, sulphur extracted from hydrocarbons and all other lease substances ("Hydrocarbons") under the Leases and that may be produced and saved under or otherwise be allocated or attributed to the Leases; (c) the oil, gas, water or injection wells located on Leases or Units, whether producing, shut-in or temporarily abandoned, including those described on Exhibit B (the "Wells") and including all of the personal property, equipment, fixtures and improvements used in connection therewith; (d) the unitization, pooling and communitization agreements, declarations, orders and the units created thereby relating to the properties and interests described in clauses (a) through (c) or to the production, gathering, treatment, processing, storage, sale, disposal and other handling of Hydrocarbons, if any,

attributable to said properties and interests; (e) all equipment, machinery, inventory, fixtures and other tangible personal property and improvements located on or used or held for use in connection with the operation of the interests described in clauses (a) through (d) or the production, gathering, treatment, processing, storage, sale, disposal, and other handling of Hydrocarbons attributable thereto, including any wells, tanks, boilers, buildings, offices (including office leases and office furniture and office supplies located therein), fixtures, injection facilities, saltwater disposal facilities, compression facilities, pumping units and engines, platforms, flow lines, pipelines, gathering systems, gas and oil treating facilities, machinery, vehicles, power lines, telephone and telegraph lines, roads, and other appurtenances, improvements and facilities, including personal computers and associated peripherals, radio, telephone, telemetry and SCADA equipment (all of the foregoing, collectively, the "Equipment"); (f) all surface leases, permits, rights-of-way, licenses, easements and other surface rights agreements used in connection with the production, gathering, treatment, processing, storage, sale, disposal and other handling of Hydrocarbons or produced water from the interests described in clauses (a) through (e) (collectively, the "Surface Contracts"); (g) all existing and effective contracts, sales and purchase contracts, operating agreements, exploration agreements, development agreements, balancing agreements, farmout agreements, service agreements, transportation, processing, treatment or gathering agreements, equipment leases and other contracts, agreements and instruments, insofar as they directly relate to the properties and interests described in clauses (a) through (f) (collectively, the "Contracts"); and (h) to the extent transferable without payment of additional consideration by SM Energy, originals, to the extent available, or copies of all the files, records and data relating to the items described in clauses (a) through (g) above, which records shall include, without limitation: lease records, well records, division order records, well files, title records (including abstracts of title, title opinions and memoranda, and title curative documents), engineering records, geological and geophysical data (including seismic data) and all technical evaluations, interpretative data and technical data and information relating to the Assets, correspondence, electronic data files (if any), maps, production records, electric logs, core data, pressure data, decline curves and graphical production curves, reserve reports, appraisals and accounting and Asset Tax records (collectively, the "Records").

"Assignment" means the Assignment and Bill of Sale from SM Energy to Buyer, pertaining to the Assets, substantially in the form attached hereto a Exhibit C.

"Assumed Obligations" has the meaning set forth in Section 13.1.

"Atokan Interval" means (A) with respect to the Broxton Area, the stratigraphic equivalent of the interval encountered between the specific marker bed in the Atoka at 17,271' and the top of the Morrow, defined as the correlative surface to the point on the log at a measured depth of 18,750', as defined by the Array Induction Tool log for the SM Energy Company Clift 1-12 (API # 35015230590000) and dated August 2008; (B) with respect to the Cleveland/Tonkawa Area, the stratigraphic equivalent of the interval encountered between the top of the Atoka at 13,161' and the base of the Atoka as defined as the correlative surface on the log at a measured depth of 13,996', as defined by the Dual Induction SFL log for the Inexco Oil Company Bradshaw 1-15 (API # 35129208280000) and dated April 1982; and (C) with respect to the

Mayfield/Sixty Six Area, the stratigraphic equivalent of the interval encountered between the top of the Atoka at 15,014' and the base of the Atoka as defined as the correlative surface on the log at a measured depth of 16,400', as defined by the Schlumberger Array Induction Image log for the SM Energy

Company (listed as St Mary Operating Company, in the well header) Cynthia 3-21 (API#35009213230000) and dated October 2003.

- "Braithwaite Area" means the Assets within the area indicated on Exhibit D-1.
- "Broxton Area" means the Assets within the area indicated on Exhibit D-2.
- "Burdens" means, with respect to any Asset, all royalties, overriding royalties, production payments, carried interests, net profits interests, reversionary interests and other burdens upon, measured by or payable out of, production therefrom.
 - "Business Day" means a day (other than a Saturday or Sunday) on which commercial banks in Texas are generally open for business.
 - "Buyer" has the meaning set forth in the preamble to this Agreement.
 - "Buyer Indemnified Parties" has the meaning set forth in Section 13.2.
 - "Buyer's Representatives" has the meaning set forth in Section 5.1(a).
 - "Casualty Loss" has the meaning set forth in Section 6.3(b).
- "Chieftain Covered Legal Expenses" means (a) all Chieftain Legal Expenses incurred or accrued prior to the Effective Time (excluding any amounts paid or payable to the plaintiffs in the Chieftain Litigation) and (b) the Reimbursable Chieftain Legal Expenses.
- "Chieftain Legal Expenses" means all costs and expenses (including fees of counsel, court costs, costs of investigation, costs of expert witnesses, mediation expenses, settlement expenses and other related costs and expenses) related to or attributable to the defense or prosecution of the Chieftain Litigation.
 - "Chieftain Litigation" means that litigation identified as item 1 on Schedule 3.6, insofar as such litigation relates to the Assets.
 - "Claim Notice" has the meaning set forth in Section 13.7(b).
- "Cleveland Interval" means (A) with respect to the Cleveland/Tonkawa Area, the stratigraphic equivalent of the interval encountered between the base of the Missourian at 10,639' and the top of the Desmoinesian defined as the correlative surface to the point on the log at a measured depth of 11,226', as defined by the Dual Induction SFL log for the Inexco Oil Company Bradshaw 1-15 (API # 35129208280000) and dated April 1982; and (B) the stratigraphic equivalent of the interval encountered between the base of the Missourian at 12,132' and the top of the Desmoinesian defined as the correlative surface to the point on the log at a measured depth of 12,683', as defined by the Schlumberger Array Induction Image log for the SM Energy Company (listed as St Mary Operating Company, in the well header) Cynthia 3-21 (API#35009213230000) and dated October 2003.
 - "Cleveland/Tonkawa Area" means the Assets within the area indicated on Exhibit D-3.
 - "Closing" has the meaning set forth in Section 11.1.
 - "Closing Amount" means the Preliminary Purchase Price less the Deposit.
 - "Closing Date" has the meaning set forth in Section 11.1.
 - "Code" means the Internal Revenue Code of 1986, as amended.
 - "Confidentiality Agreement" means that certain Confidentiality Agreement between SM Energy and Buyer, dated as of August 21, 2013.
 - "Consents" has the meaning set forth in Section 6.4.
 - "Contract" has the meaning set forth in the definition of "Assets" above.
- "Controf" and its derivatives mean, with respect to any Person, the possession, directly or indirectly, of the power to exercise or determine the voting of more than 50% of the voting rights in a corporation, and, in the case of any other type of entity, the right to exercise or determine the voting of more than 50% of the equity interests having voting rights, or otherwise to direct or cause the direction of the management and policies of such Person, whether by contract or otherwise.
 - "Cure Assets" has the meaning set forth in Section 6.2(c).
 - "Cure Period" has the meaning set forth in Section 6.2(c)(ii).
- "Customary Post-Closing Consents" means those consents and approvals from Governmental Authorities for the assignment of the Assets to the Buyer that are customarily obtained after the assignment of properties similar to the Assets.
- "Deep Interval" means (A) with respect to the Braithwaite Area, the stratigraphic equivalent of the interval encountered from the top of the Atoka at 13,725' to all correlative surfaces that occur deeper than this point, as defined by the Schlumberger Dual Induction SFL log for SM Energy Company (listed as CHK Company in the well header) Weter 1-35 (API# 35149202280000), spud August 1981 (no date on log header); (B) with respect to the Cleveland/Tonkawa Area, the stratigraphic equivalent of the interval encountered from the base of the Atoka at 13,996' to all correlative surfaces that occur deeper than this point, as defined by the Dual Induction SFL log for the Inexco Oil Company Bradshaw 1-15 (API # 35129208280000) and dated April 1982; (C) with respect to the Mayfield/Sixty Six Area, the stratigraphic equivalent of the interval encountered from the base of the Atoka at 16,400' to all correlative surfaces that occur deeper than this point, as defined by the Schlumberger Array Induction Image log for the SM Energy Company (listed as St Mary Operating Company, in the well header) Cynthia 3-21 (API#35009213230000) and dated October 2003; and (D) with respect to the Woodford Area, the stratigraphic equivalent of all correlative surfaces that occur deeper than the top of the Hunton Limestone as defined on the log at a depth of 11,410', as defined by the Schlumberger Array Induction Imager log for SM Energy Company (listed as St Mary Operating Company, in the well header) Good Farms 1-27 (API# 35051229480000) and dated February 2001.
- "Defect Claim Date" has the meaning set forth in Section 6.2(a).

"Defensible Title" means such title of SM Energy with respect to the Assets that, subject to Permitted Encumbrances:

- (a) with respect to each Well or Well Location shown in Schedule 2.5 (but limited to the applicable Target Interval set forth on Schedule 2.5 for such Well or Well Location and any currently producing formation (for such Well)), entitles SM Energy to receive not less than the Net Revenue Interest shown in Schedule 2.5, for such Well or Well Location, except for (i) decreases in connection with those operations in which SM Energy may, from and after the Execution Date, be a non-consenting co-owner (to the extent permitted pursuant to Section 8.1),
- (ii) decreases resulting from the establishment or amendment from and after the Execution Date of pools or units, (iii) decreases required to allow other Working Interest owners to make up past underproduction or pipelines to make up past under deliveries, to the extent reflected on Schedule 3.10, and (iv) as otherwise stated in Schedule 2.5;
- (b) with respect to each Well or Well Location shown in Schedule 2.5 (but limited to the applicable Target Interval set forth on Schedule 2.5 for such Well or Well Location and any currently producing formation (for such Well)), obligates SM Energy to bear the Working Interest for such Well or Well Location not greater than the Working Interest shown in Schedule 2.5, for such Asset, except (i) increases resulting from contribution requirements with respect to defaults by co-owners from and after the Execution Date under Applicable Operating Agreements, (ii) increases to the extent that they are accompanied by a proportionate increase in the Net Revenue Interest in the Asset, and (iii) as otherwise stated in Schedule 2.5; and
 - (c) is free and clear of all Encumbrances.
 - "Deposit" has the meaning set forth in Section 2.2(a).
- "Desmoines Interval" means (A) with respect to the Braithwaite Area, the stratigraphic equivalent of the interval encountered between the top of the Desmoines at 12,043' and the top of the Atoka, defined as the correlative surface to the point on the log at a measured depth of 13,725', as defined by the Schlumberger Dual Induction SFL log for SM Energy Company (listed as CHK Company in the well header) Weter 1-35 (API# 35149202280000), spud August 1981 (no date on log header); (B) with respect to the Cleveland/Tonkawa Area, the stratigraphic equivalent of the interval encountered between the top of the Desmoinesian at 11,226' and the top of the Atokan as defined as the correlative surface to the point on the log at a measured depth of 13,161', as defined by the Dual Induction SFL log for the Inexco Oil Company Bradshaw 1-15 (API # 35129208280000) and dated April 1982; and (C) with respect to the Mayfield/Sixty Six Area, the stratigraphic equivalent of the interval encountered between the top of the Desmoinesian at 12,683' and the top of the Atokan as defined as the correlative surface to the point on the log at a measured depth of 15,014', as defined by the Schlumberger Array Induction Image log for the SM Energy Company (listed as St Mary Operating Company, in the well header) Cynthia 3-21 (API#35009213230000) and dated October 2003.
 - "Digital Records" has the meaning set forth in Section 8.5.
- "Dispute" means any dispute, controversy or claim (of any and every kind or type, whether based on contract, tort, statute, regulation or otherwise) arising out of, relating to or connected with this Agreement or the transactions contemplated hereby, including but not limited to any dispute, controversy or claim concerning the existence, validity, interpretation, performance, breach or termination of this Agreement, the relationship of the Parties arising out of this Agreement or the transactions contemplated hereby.
 - "Dispute Notice" has the meaning set forth in Section 2.6(b).
 - "Effective Time" means 7:00 a.m. local time at the location of the Assets on October 1, 2013.
- "Encumbrance" means a mortgage, lien, security interest, pledge, charge or other encumbrance, and "Encumber" and other similar derivatives shall be construed accordingly.
 - "Environmental Arbitrator" has the meaning set forth in Section 7.1(e).
- "Environmental Condition" means (a) a condition existing on the Defect Claim Date with respect to the air, soil, subsurface, surface waters, ground waters and/or sediments that causes any Asset (or SM Energy with respect to any Asset) not to be in compliance with any Environmental Law or (b) the existence as of the Execution Date with respect to any Asset or the operation thereof of any environmental pollution, contamination, degradation, damage or injury caused by, related to such Asset for which remedial or corrective action is presently required (or if known, would be presently required) under Environmental Laws.
 - "Environmental Defect" means an Environmental Condition with respect to an Asset that is not set forth in Schedule 3.12.
 - "Environmental Defect Notice" has the meaning set forth in Section 7.1(a).
 - "Environmental Defect Property" has the meaning set forth in Section 7.1(a).
- "Environmental Laws" means all applicable federal, state and local Laws in effect as of the Execution Date, including common Law, relating to the protection of the public health, welfare and the environment, including, without limitation, those Laws relating to the generation, storage, handling, use, processing, treatment, transportation, disposal or other management of chemicals and other Hazardous Substances. The term "Environmental Laws" does not include good or desirable operating practices or standards that may be employed or adopted by other oil and gas well operators or recommended by any Governmental Authority.
 - "Equipment" has the meaning set forth in the definition of "Assets" above.
 - "Escrow Account" has the meaning set forth in Section 2.2.
 - "Escrow Agent" has the meaning set forth in Section 2.2.
 - "Escrow Agreement" has the meaning set forth in Section 2.2.
 - "Execution Date" has the meaning set forth in the preamble to this Agreement.
- "Excluded Assets" means (a) all of SM Energy's corporate minute books, financial records and other business records that relate to SM Energy's business generally (including the ownership and operation of the Assets); (b) all trade credits, all accounts, receivables and all other proceeds, income or revenues attributable to the Assets with respect to any period of time prior to the Effective Time; (c) subject to Section 6.3, all rights and interests relating to the Assets (i) under any existing policy or agreement of insurance, (ii) under any bond or (iii) to any insurance or condemnation proceeds or awards arising, in each case, from acts, omissions or events, or damage to or destruction of property; (d) all Hydrocarbons produced and sold from the Assets with respect to all periods prior to the Effective Time; (e) all claims of SM Energy or its Affiliates for refunds of or loss carry forwards with respect to (i) production or any other Taxes paid by SM Energy or its Affiliates attributable to any period prior to the Effective Time,

(ii) income Taxes paid by SM Energy or its Affiliates or (iii) any Taxes attributable to the other Excluded Assets; (f) all of SM Energy's proprietary computer software, patents, trade secrets, copyrights, names, trademarks, logos and other intellectual property; (g) all documents and instruments of SM Energy that may be protected by an attorney-client privilege (other than title opinions and reports on status of title, in each case, with respect to title to any of the Assets); (h) all data that cannot be disclosed to Buyer as a result of confidentiality arrangements under agreements with third parties; (i) all audit rights arising under any of the (i) Applicable Contracts or otherwise with respect to any period prior to the Effective Time or (ii) other Excluded Assets,

except for any Imbalances; (j) all geophysical and other seismic and related technical data and information relating to the Assets to the extent that such geophysical and other seismic and related technical data and information is not transferable without payment of a fee or other penalty to any third party under any Contract and which Buyer has not separately agreed in writing to pay; (k) documents prepared or received by SM Energy or its Affiliates with respect to (i) lists of prospective purchasers for the Assets, (ii) bids submitted by other prospective purchasers of the Assets, (iii) analyses by SM Energy or its Affiliates of any bids submitted by any prospective purchaser, (iv) correspondence between or among SM Energy, its representatives and any prospective purchaser other than Buyer, and (v) correspondence between SM Energy or any of its representatives with respect to any of the bids, the prospective purchasers or the transactions contemplated by this Agreement; (l) a copy of all Records; (m) any Applicable Contracts and Records that are related to Assets that are excluded pursuant to the provisions of Section 6.4(c)(ii), Section 6.4(d)(i) or Section 7.1(b)(ii); (n) any Contracts that constitute master services agreements or similar contracts; and (o) all assets listed on Exhibit F.

"Final Settlement Statement" has the meaning set forth in Section 2.6(b).

"Franchise Tax Liability" means any Tax imposed by a state on SM Energy's or any of its Affiliates' gross or net income and/or capital for the privilege of engaging in business in that state that was or is attributable to SM Energy's or any of its Affiliates' ownership of an interest in the Assets.

"GAAP" means the generally accepted accounting principles in the United States of America.

"Governmental Authority" means any federal, state, local, municipal, tribal or other government; any governmental, regulatory or administrative agency, commission, body or other authority exercising or entitle to exercise any administrative, executive, judicial, legislative, regulatory or taxing authority or power; and any court or governmental tribunal, including any tribal authority having or asserting jurisdiction.

"Hard Consent" means any Consent other than a Soft Consent.

"Hazardous Substances" means any pollutants, contaminants, toxics or hazardous or extremely hazardous substances, materials, wastes, constituents, compounds or chemicals that are regulated by, or may form the basis of any liability under, any Environmental Laws, including NORM and other substances referenced in Section 7.2.

"Hydrocarbons" has the meaning set forth in the definition of "Assets" above.

"Imbalance" means any imbalance at the (a) wellhead between the amount of Hydrocarbons produced from a Well and allocable to the interests of SM Energy therein and the shares of production from the relevant Well to which SM Energy is entitled or (b) pipeline flange between the amount of Hydrocarbons nominated by or allocated to SM Energy and the Hydrocarbons actually delivered on behalf of SM Energy at that point.

"Income Tax Liability" means any Liability of SM Energy or any of its Affiliates attributable to any federal, state or local income Tax measured by or imposed on the income of SM Energy or any of its Affiliates that was or is attributable to SM Energy's or any of its Affiliates' ownership of an interest in or the operation of the Assets.

"Indemnified Party" has the meaning set forth in Section 13.7(a).

"Indemnifying Party" has the meaning set forth in Section 13.7(a).

"Individual Environmental Threshold" means [***].

"Individual Title Defect Threshold" means [***].

"Interim Period" means that period of time from and after the Effective Time up to Closing.

"Invasive Activities" has the meaning set forth in Section 5.1(b).

"Knowledge" means with respect to SM Energy, the actual knowledge (after reasonable inquiry) of the following Persons: Mary Ellen Lutey, Vice President and Regional Manager, Kenneth Knott, Vice President — Land, Kevin Wilson, Land Manager, Hans Swolfs, Landman, John Rosata, Regional EHS Manager, Randy Buckley, Asset Team Manager, and Marlon Wells, Regional Operations Manager.

"Laws" means any constitution, decree, resolution, law, statute, act, ordinance, rule, directive, order, treaty, code or regulation and any injunction or final non-appealable judgment or any interpretation of the foregoing, as enacted, issued or promulgated by any Governmental Authority.

"Leases" has the meaning set forth in the definition of "Assets" above.

"Liabilities" means any and all claims, causes of actions, payments, charges, judgments, assessments, liabilities, losses, damages, penalties, fines, costs and expenses, including any attorneys' fees and legal or other expenses incurred in connection therewith and including liabilities, costs, losses and damages for personal injury or death or property damage.

"Material Adverse Effect" means any change, inaccuracy, effect, event, result, occurrence, condition or fact (for the purposes of this definition, each, an "event") (whether foreseeable or not and whether covered by insurance or not) that has had or would be reasonably likely to have, individually or in the aggregate with any other event or events, a material adverse effect on the ownership, operation or financial condition of an Asset; provided, however, that "Material Adverse Effect" shall not include such material adverse effects resulting from (a) general changes in oil and gas prices; (b) general changes in industry, economic or political conditions or markets; (c) changes in conditions or developments generally applicable to the oil and gas industry in Texas or Oklahoma; (d) acts of God, including hurricanes and storms; (e) acts or failures to act of Governmental Authorities; (f) civil unrest or similar disorder, terrorist acts or changes in Laws; (g) effects or changes that are cured or no longer exist by the earlier of Closing and the termination of this Agreement pursuant to Section 12.1, without cost to Buyer; (h) changes in GAAP; and (i) changes resulting from the announcement of the transactions contemplated hereby or the performance of the covenants set forth in Article VIII; provided that, in each case, the changes and effects described in clauses (a), (b) and (c) of this definition do not disproportionately affect the Assets, taken as a whole.

"Material Contracts" has the meaning set forth in Section 3.7(a).

"Mayfield/Sixty Six Area" means the Assets within the area indicated on Exhibit D-4.

"Middle Interval" means, with respect to the Braithwaite Area, the stratigraphic equivalent of the interval encountered between the specific marker bed in the Upper Virgilian at 8,339' and the top of the Desmoines, defined as the correlative surface to the point on the log at a measured depth of 12,043', as defined by the Schlumberger Dual Induction SFL log for SM Energy Company (listed as CHK Company in the well header) Weter 1-35 (API# 35149202280000), spud August 1981 (no date on log header).

"Missourian Interval" means (A) with respect to the Cleveland/Tonkawa Area, the stratigraphic equivalent of the interval encountered between the top of the Missourian at 9,412' and the base of the Missourian defined as the correlative surface to the point on the log at a measured depth of 10,639', as defined by the Dual Induction SFL log for the Inexco Oil Company Bradshaw 1-15 (API # 35129208280000) and dated April 1982; and (B) with respect to the Mayfield/Sixty Six Area, the stratigraphic equivalent of the interval encountered between the top of the Missourian at 10,689' and the base of the Missourian defined as the correlative surface to the point on the log at a measured depth of 12,132', as defined by the Schlumberger Array Induction Image log for the SM Energy Company (listed as St Mary Operating Company, in the well header) Cynthia 3-21 (API#35009213230000) and dated October 2003.

"Month" means any of the months of the Gregorian calendar.

"Morrow Interval" means, with respect to the Broxton Area, the stratigraphic equivalent of the interval encountered between the top of the Morrow at 18,750' and the top of the Springer defined as the correlative surface to the point on the log at a measured depth of 19,962', as defined by the Array Induction Tool log for the SM Energy Company Clift 1-12 (API # 35015230590000) and dated August 2008.

"Net Revenue Interest" means, with respect to any Well or Well Location (but limited to the applicable Target Interval set forth on Schedule 2.5 for such Well or Well Location and any currently producing formation (for such Well)), the interest in and to all Hydrocarbons produced, saved and sold from or allocated to such Well or Well Location (but limited to the applicable Target Interval set forth on Schedule 2.5 for such Well or Well Location and any currently producing formation (for such Well)), after giving effect to all Burdens thereon.

"NORM" means naturally occurring radioactive material.

"Operating Expenses" means all operating expenses (including Asset Taxes) and capital expenditures incurred in the ownership and operation of the Assets in the ordinary course of business and, where applicable, in accordance with the Applicable Operating Agreement, and overhead costs charged to the Assets under the Applicable Operating Agreement, but excluding Liabilities attributable to (a) personal injury or death, property damage or violation of any Law, (b) obligations to plug wells and dismantle or decommission facilities, (c) the Remediation of any Environmental Condition under applicable Environmental Laws, (d) obligations with respect to Imbalances, or (e) obligations to pay Working Interests, royalties, overriding royalties or other interest owners revenues or proceeds attributable to sales of Hydrocarbons relating to the Assets, including those held in suspense.

"Operator" means the Person serving as operator under any Applicable Operating Agreement.

"Outside Termination Date" has the meaning set forth in Section 12.1(a).

"Party" and "Parties" have the meanings set forth in the preamble to this Agreement.

"Permitted Encumbrances" means:

(a) lessor's royalties, non-participating royalties, overriding royalties, reversionary interests and similar burdens upon, measured by or payable out of production if the net cumulative effect of such burdens does not (i) operate to reduce the Net Revenue Interest of SM Energy in any Well or Well Location to an amount less than the Net Revenue Interest set forth on *Schedule 2.5* for such Well or Well Location, (ii) obligate SM Energy to bear a Working

Interest for such Well or Well Location in any amount greater than the Working Interest set forth on Schedule 2.5 for such Well or Well Location (unless the Net Revenue Interest for such Well or Well Location is greater than the Net Revenue Interest set forth on Schedule 2.5, in the same proportion as any increase in such Working Interest) or (iii) increase the Burdens for any Well or Well Location to an amount greater than that set forth on Schedule 2.5;

- (b) Preferential Rights or similar agreements with respect to which (A) waivers are obtained from the appropriate parties for the transaction contemplated hereby, or (B) required notices have been given for the transaction contemplated hereby to the holders of such rights and the appropriate period for making an election has expired without an exercise of such rights;
- (c) required third party consents to assignments or similar agreements with respect to which (A) consents have been obtained from the appropriate parties for the transaction contemplated hereby, or (B) required notices have been given for the transaction contemplated hereby to the holders of such rights and the applicable period (as specified in the contract, agreement or other instrument granting or reserving such rights) for giving notice of objection or withholding of consent has expired without an exercise of such rights or the period within which the failure to respond to such notice is considered under the relevant contract, agreement or other instrument as deemed consent has expired without SM Energy's receipt of a notice of objection or withholding of consent;
 - (d) liens for Taxes or assessments not yet due or delinquent;
 - (e) Customary Post-Closing Consents;
 - (f) other than such rights that have already been triggered, conventional rights of reassignment;
- (g) such Title Defects as Buyer may have waived or is deemed to have waived pursuant to the terms of this Agreement or Title Defects that were not properly asserted by Buyer prior to the Defect Claim Date (other than claims which may be made pursuant to the special warranty of title set forth in the Assignment);
- (h) all applicable Laws, and rights reserved to or vested in any Governmental Authority (i) to control or regulate any Asset in any manner; (ii) by the terms of any right, power, franchise, grant, license or permit, or by any provision of Law, to terminate such right, power, franchise grant, license or permit or to purchase, condemn, expropriate or recapture or to designate a purchaser of any of the Assets; (iii) to use such property in a manner which does not materially impair the use of such property for the purposes for which it is currently owned and operated and (iv) to enforce any obligations or duties affecting the Assets to any Governmental Authority, with respect to any franchise, grant, license or permit;
- (i) rights of a common owner of any interest in rights-of-way or easements currently held by SM Energy and such common owner as tenants in common or through common ownership to the extent that the same does not materially impair the use or operation of the Assets as currently used and operated;
 - (j) easements, conditions, covenants, restrictions, servitudes, permits, rights-of-way, surface leases and other rights in the Assets for the purpose of

surface operations, roads, alleys, highways, railways, pipelines, transmission lines, transportation lines, distribution lines, power lines, telephone lines and removal of timber, grazing, logging operations, canals, ditches,

reservoirs and other like purposes, or for the joint or common use of real estate, rights-of-way, facilities and equipment, (in each case) that do not (i) materially impair the use, ownership or operation of the Assets (as currently owned and operated), (ii) reduce the Net Revenue Interest of SM Energy in any Well or Well Location to an amount less than the Net Revenue Interest set forth on *Schedule 2.5*, for such Well or Well Location, or (iii) obligate SM Energy to bear a Working Interest for such Well or Well Location in any amount greater than the Working Interest set forth on *Schedule 2.5* for such Well or Well Location (unless the Net Revenue Interest for such Well or Well Location is greater than the Net Revenue Interest set forth on *Schedule 2.5*, in the same proportion as any increase in such Working Interest) or increase the Burdens for any Well or Well Location to an amount greater than that set forth on *Schedule 2.5*;

- (k) zoning and planning ordinances and municipal regulations;
- (l) vendors, carriers, warehousemen's, repairmen's, mechanics, workmen's, materialmen's, construction or other like Encumbrances arising by operation of Law in the ordinary course of business or incident to the construction or improvement of any property in respect of obligations that are not yet due;
 - (m) Encumbrances created under Leases and/or Applicable Operating Agreements or by operation of Law in respect of obligations that are not yet due;
 - (n) any Encumbrance affecting the Assets which is discharged by SM Energy at or prior to Closing;
 - (o) any Contracts, restrictions or exclusions set forth on **Exhibit A** or **Exhibit B**, as applicable, and all litigation referenced in *Schedule 3.6*; and
- (p) the Leases and all other Encumbrances, Contracts (including the Applicable Contracts), agreements, instruments, obligations, defects and irregularities affecting the Assets that (in each case) do not (i) materially impair the use, ownership or operation of the Assets (as currently owned and operated), (ii) reduce the Net Revenue Interest of SM Energy in any Well or Well Location to an amount less than the Net Revenue Interest set forth on *Schedule 2.5* for such Well or Well Location, (iii) obligate SM Energy to bear a Working Interest for such Well or Well Location in any amount greater than the Working Interest set forth on *Schedule 2.5* for such Well or Well Location (unless the Net Revenue Interest for such Well or Well Location is greater than the Net Revenue Interest set forth on *Schedule 2.5*, in the same proportion as any increase in such Working Interest), or (iv) increase the Burdens for any Well or Well Location to an amount greater than that set forth on *Schedule 2.5*.
- "Person" means any individual, firm, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization, Governmental Authority or any other entity.
- "Personal Property" means equipment, machinery, fixtures, and other real, immovable, personal, movable and mixed property, including saltwater disposal wells, well equipment, casing, rods, tanks, boilers, buildings, tubing, pumps, motors, fixtures, machinery, compression equipment, flow lines, and separation facilities, structures, materials, and other items used or held for use in the operation thereof and located upstream of the outlet flange of the relevant custody transfer meter (or, in the case of Hydrocarbon liquids, upstream of the outlet flange in the tanks).
 - "Preferential Right" has the meaning set forth in Section 6.4.
 - "Preliminary Settlement Statement" has the meaning set forth in Section 2.6(a).
 - "Proportionate Share" has the meaning set forth in the Recitals.
 - "Purchase Price" has the meaning set forth in Section 2.2, as such amount may be adjusted from time to time pursuant to Section 2.4 and Section 2.6.
 - "Records" has the meaning set forth in the definition of "Assets" above.
- "Reimbursable Chieftain Legal Expenses" means the first Three Million Dollars (\$3,000,000) of Chieftain Legal Expenses incurred or accrued on or after the Effective Time.
 - "Reimbursement Obligation" has the meaning set forth in Section 8.6.
- "Remediation" means, with respect to an Environmental Condition, the implementation and completion, in a manner agreed to by the Parties, of any remedial, removal, response, construction, closure, disposal or other corrective actions required under Environmental Laws to correct or remove such Environmental Condition.
- "Remediation Amount" means, with respect to an Environmental Condition, the present value as of the Closing Date of the cost (net to the Asset) of the most cost effective Remediation of such Environmental Condition, as agreed to by the Parties, that is reasonably available, plus any and all fines, penalties, court and/or administrative costs and expenses and all other costs and expenses (including Liabilities to third parties) attributable to or arising from the existence of such Environmental Condition.
- "Schedules" means the schedule delivered to Buyer prior to the execution of this Agreement setting forth specific exceptions to SM Energy's representations and warranties set forth in this Agreement.
- "Shallow Interval" means (A) with respect to the Braithwaite Area, the stratigraphic equivalent of the interval encountered between the surface of the Earth and the correlative point on the log corresponding to a marker bed with a measured depth of 8,339' in the Upper Virgilian, as defined by the Schlumberger Dual Induction SFL log for SM Energy Company (listed as CHK Company in the well header) Weter 1-35 (API# 35149202280000), spud August 1981 (no date on log header); (B) with respect to the Broxton Area, the stratigraphic equivalent of the interval encountered between the surface of the Earth and the correlative point on the log corresponding to a marker bed at the top of the Atoka with a measured depth of 17,271', as defined by the Array Induction Tool log for the SM Energy Company Clift 1-12 (API # 35015230590000) and dated August 2008; (C) with respect to the Cleveland/Tonkawa Area, the stratigraphic equivalent of the interval encountered between the surface of the Earth and the correlative point on the log corresponding to a marker bed with a measured depth of 8,015' in the Upper Virgilian, as defined by the Dual Induction SFL log for the Inexco Oil Company Bradshaw 1-15 (API # 35129208280000) and dated April 1982; (D) with respect to the Mayfield/Sixty Six Area, the stratigraphic equivalent of the interval encountered between the surface of the Earth and the correlative point on the log corresponding to a marker bed with a measured depth of 9,102' in the Upper Virgilian, as defined by the Schlumberger Array Induction Image log for the SM Energy Company (listed as St Mary Operating Company, in the well header) Cynthia 3-21 (API#35009213230000) and dated October 2003; and (E) with respect to the Woodford Area, the stratigraphic equivalent of the interval encountered between the surface of the Earth and the top of the Woodford shale defined as a point on the log corresponding to a measured depth of 11,276', as defined by the Schlumberger Array Induction

Imager log for SM Energy Company (listed as St Mary Operating Company, in the well header) Good Farms 1-27 (API# 35051229480000) and dated February 2001.

- "SM Energy" has the meaning set forth in the preamble to this Agreement.
- "SM Indemnified Parties" has the meaning set forth in Section 13.3.
- "Soft Consent" means a Consent that by its terms may not be unreasonably withheld, conditioned or delayed, and for which the failure to obtain would not (i) cause the assignment to Buyer of any portion of the Assets to be void, (ii) cause the termination of a Lease or Contract under the express terms thereof, or (iii) result in a penalty of any sort, including a requirement to make a payment.
- "Springer Interval" means, with respect to the Broxton Area, the stratigraphic equivalent of the interval encountered from the base of the top of the Springer at 19,962' to all correlative surfaces deeper than this point, as defined by the Array Induction Tool log for the SM Energy Company Clift 1-12 (API # 35015230590000) and dated August 2008.
 - "Subject Transfer Taxes" has the meaning set forth in Section 10.2.
 - "Surface Contracts" has the meaning set forth in the definition of "Assets" above.
- "Target Interval," means the Atokan Interval, the Cleveland Interval, the Desmoines Interval, the Deep Interval, the Middle Interval, the Missourian Interval, the Morrow Interval, the Shallow Interval, the Springer Interval, the Virgilian Interval and the Woodford Interval, as applicable.
- "Taxes" means any and all federal, state, local, foreign and other taxes or other assessments, including, without limitation, all net income, gross income, gross receipts, sales, use, ad valorem, transfer, franchise, profits, profit share, license, lease, service, service use, value added, withholding, payroll, employment, excise, estimated severance, stamp, occupation, premium, property, windfall profit or other taxes of any kind whatsoever, together with any interests, penalties, additions to tax, fines or other additional amounts imposed thereon or related thereto, and the term "Tax" means any one of the foregoing Taxes.
 - "Third Party Claim" has the meaning set forth in Section 13.7(b).
 - "Third Party Operator" has the meaning set forth in Section 5.1(a).
 - "Title Arbitrator" has the meaning set forth in Section 6.2(i).
- "Title Benefit" means any right, circumstance or condition that operates to increase the Net Revenue Interest being assigned to Buyer in any Well or Well Location above that shown for such Well or Well Location in Schedule 2.5, to the extent the same does not cause a greater than proportionate increase in the Working Interest being assigned to Buyer therein above that shown in Schedule 2.5.
 - "Title Benefit Amount" has the meaning set forth in Section 6.2(d).
 - "Title Benefit Notice" has the meaning set forth in Section 6.2(b).
 - "Title Benefit Property" has the meaning set forth in Section 6.2(b).
- "Title Defect" means any Encumbrance, defect or other matter that causes SM Energy not to have Defensible Title in and to any Asset; provided that the following shall not be considered Title Defects:
- (a) defects in the chain of title consisting of the failure to recite marital status in a document or omissions of successions of heirship or estate proceedings, unless Buyer provides affirmative evidence that such failure or omission could reasonably be expected to result in another Person's superior claim of title to the relevant Asset;
 - (b) defects arising out of lack of survey, unless a survey is expressly required by applicable Laws;
- (c) liens created under deeds of trust, mortgages and similar instruments by the lessor under a Lease covering the lessor's surface and mineral interests in the land covered thereby that would customarily be accepted in taking oil and gas leases or purchasing undeveloped oil and gas leases and for which the lessee would not customarily seek a subordination of such lien to the oil and gas leasehold estate prior to conducting drilling activities on the Lease;
- (d) defects arising out of lack of corporate or other entity authorization unless Buyer provides affirmative evidence that causes Buyer to reasonably believe such corporate or other entity action may not have been authorized and could reasonably be expected to result in another Person's superior claim of title to the relevant Asset;
- (e) defects based on a gap in SM Energy's chain of title in the state's records as to state Leases, or in the county records as to other Leases, unless such gap is affirmatively shown to exist in such records by an abstract of title, title opinion or landman's title chain or runsheet, which documents shall be included in a Title Defect Notice;
 - (f) defects that have been cured by applicable Laws of limitations or prescription; and
 - (g) any Encumbrance or loss of title resulting from SM Energy's conduct of business after the Execution Date in compliance with Section 8.1.
 - "Title Defect Amount" has the meaning set forth in Section 6.2(f).
 - "Title Defect Notice" has the meaning set forth in Section 6.2(a).
 - "Title Defect Property" has the meaning set forth in Section 6.2(a).
 - "Title Disputed Matters" has the meaning set forth in Section 6.2(i).
- "Treasury Regulations" means the regulations promulgated by the United States Department of the Treasury pursuant to and in respect of provisions of the Internal Revenue Code. All references herein to sections of the Treasury Regulations shall include any corresponding provision or provisions of succeeding, similar, substitute, proposed or final Treasury Regulations.
 - "Unit" has the meaning set forth in the definition of "Assets" above.

"Virgilian Interval" means (A) with respect to the Cleveland/Tonkawa Area, the stratigraphic equivalent of the interval encountered between the specific marker bed in the Upper Virgilian at 8,015' and the top of the Missourian, defined as the correlative surface to the point on the log at a measured depth of 9,412', as defined by the Dual Induction SFL log for the Inexco Oil Company Bradshaw 1-15 (API # 35129208280000) and dated April 1982; and (B) with respect to the Mayfield/Sixty Six Area, the stratigraphic equivalent of the interval encountered between the specific marker bed in the Upper Virgilian at 9,102' and the top of the Missourian, defined as

the correlative surface to the point on the log at a measured depth of 10,689°, as defined by the Schlumberger Array Induction Image log for the SM Energy Company (listed as St Mary Operating Company, in the well header) Cynthia 3-21 (API#35009213230000) and dated October 2003.

"Well" has the meaning set forth in the definition of "Assets" above.

"Well Location" means each well location and the assumed 640 acre spacing unit or governmental spacing unit associated with such well location, as set forth or Schedule 2.5.

"Woodford Area" means the Assets within the area indicated on Exhibit D-5.

"Woodford Interval" means, with respect to the Woodford Area, the stratigraphic equivalent of the interval encountered between the top of the Woodford Shale at 11,276' to the top of the Hunton Limestone, defined as the correlative surface to the point on the log at a measured depth of 11,410', as defined by the Schlumberger Array Induction Imager log for SM Energy Company (listed as St Mary Operating Company, in the well header) Good Farms 1-27 (API# 35051229480000) and dated February 2001.

"Working Interest" means, with respect to any Well or Well Location (but limited to the applicable Target Interval set forth or Schedule 2.5 for such Well or Well Location and any currently producing formation (for such Well)), the interest in and to such Target Interval for such Asset or producing formation (for such Asset) that is burdened with the obligation to bear and pay costs and expenses of maintenance, development and operations on or in connection with such Target Interval for such Asset or producing formation (for such Asset), but without regard to the effect of any Burdens.

EXHIBIT E

Buyer's Proportionate Shares

EnerVest Energy Institutional Fund XIII-A, L.P.	43.37281 %
EnerVest Energy Institutional Fund XIII-WIB, L.P.	1.44651%
EnerVest Energy Institutional Fund XIII-WIC, L.P.	55.18068%

CERTIFICATION

I, Anthony J. Best, certify that:

- 1. I have reviewed this Amendment No. 1 on Form 10-K/A of SM Energy Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2014

/s/ Anthony J. Best
Anthony J. Best
Chief Executive Officer

CERTIFICATION

I, A. Wade Pursell, certify that:

- 1. I have reviewed this Amendment No. 1 on Form 10-K/A of SM Energy Company;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2014

/s/ A. Wade Pursell

A. Wade Pursell

Executive Vice President and Chief Financial Officer